BERJ BERA

Religion as a barrier to the use of student loans for higher education: A community-based participatory study with Somalis living in England

Alija Avdukic¹ | Fawad Khaleel² | Ahmed Abdullah³ | Abdulbasit H. Brawe⁴

¹Al-Maktoum College of Higher Education and School of Business, University of Dundee, Dundee, UK

²The Business School, Edinburgh Napier University, Craiglockhart Campus, Edinburgh, UK

³Graduate School, Abertay University, Dundee, UK

⁴School of Business – University of Dundee, Dundee, UK

Correspondence

Alija Avdukic, Al-Maktoum College of Higher Education and School of Business University of Dundee, 124 Blackness Road, Dundee DD1 5PE, UK. Email: a.avdukic@almcollege.ac.uk; aavdukic001@dundee.ac.uk

Abstract

The unwillingness of the Somali community to finance higher education has largely gone unnoticed within the academic literature and government policy documents. This study explores the role of religion and the influence of Shari'ah scholars on the use of interest-bearing student loans within the Somali community. In the absence of any theoretical framework on this topic, we explore the multiple socioeconomic factors that may influence the attitude, perception of need, motivation and action of using student loans for higher education, by proposing the UK Somali Muslims Acceptance of Interest-bearing Student Loan Model. This is also a community-based participatory study that actively involved Somali community members in exploring and interpreting the results. This was achieved through regular consultations with the sampled Somali Muslim communities within the UK. Our results contribute to the broader debate on the effect of cultural, religious and social values of marginalised communities on inclusion and widening access policies for higher education. The findings reemphasise that people sharing the same location do not necessarily share the same level of opportunities for higher education because of the intersectionality of race, religion, gender and class.

This is an open access article under the terms of the Creative Commons Attribution-NonCommercial-NoDerivs License, which permits use and distribution in any medium, provided the original work is properly cited, the use is non-commercial and no modifications or adaptations are made.

© 2022 The Authors. British Educational Research Journal published by John Wiley & Sons Ltd on behalf of British Educational Research Association.

1

The results also show the complexity of the issue of exclusion and the atheoretical nature of student loans as a financial instrument for improving financial inclusion and widening access to higher education among Somali residents in England.

KEYWORDS

ethical religious values, Somali Muslim community, student loans, widening participation

Key insights

What is the main issue that the paper addresses?

This study explores the relationship between the religious, cultural and social values of Somali Muslim communities that impose barriers to the use of interest-bearing student loans, while exploring their perceptions and attitudes towards higher education and the wider factors that influence their decisions on getting access to higher education in England.

What are the main insights that the paper provides?

The empirical findings of this community-based participatory research show that Somali students are generally aware of religious tolerance towards the use of student loans; however the decision of using student loans (or not) is mostly intersectioned by community, identity and culture, which differ across gender and socioeconomic classes.

INTRODUCTION

The financial instrument of student loans is purposely designed by the Government to widen access to higher education (ONS, 2020). However, the impact of student loans on the financial inclusion, financial literacy and financial capacity of underrepresented communities was not truly understood by the government over the period 2010–2021 (Rowlingson & Mckay, 2017). This is critical because if the policy on student loans is resulting in the financial exclusion of underrepresented communities, then these communities are essentially barred from higher education.

The policy papers either focus on the implications of student loan debt (Bachan, 2014; Esson & Ertl, 2016; Wakeling et al., 2017) or they attempt to compute the right proportion of student loan debt financially feasible for going to university (Loonin & Morgan, 2019). The current policy papers also discuss the political and economic feasibility of student loans on the grounds of providing higher education opportunities to low-income deprived communities across England (Czarnecki et al., 2020). The groups most often cited by the policy papers to benefit from student loans are those at the bottom of the income strata, which mostly consist of ethnic, religious and cultural minorities in England (Salmi & D'Addio, 2021).

The implementation process of evaluating the political, financial and economic feasibility of the student loan model in England was conducted using a classical top-down approach. According to this approach, the policy was constructed by high-level government officials without much consultation. Public higher education institutions were left to implement this plocy, without any freedom to adjust the student loan framework to the needs of their local communities (Jungblut et al., 2020). The academic research extensively shows the continuous underrepresentation of students from low-income/wealth divisions, despite having a government facility to acquire student loans (Findlay & Hermannsson, 2019). Several studies show that the level of financial literacy and financial capability for making financial decisions among the students has a direct impact on their perception of the affordability of higher education, particularly within the lower socioeconomic group (Eichelberger et al., 2017; McDonough & Calderone, 2006; University Studies Council, 2015). The data also show that social class (McCoy & Smyth, 2011), economic profile (HEA, 2017), social identity (Kena et al., 2015) and culture (Eichelberger et al., 2017) influence the decision to use student loans to gain access to higher education participation. However, relatively few research studies have been conducted on the role played by social identity, culture and religion of the underrepresented students in their decision to seek student loans to pursue higher education.

Therefore, this study fills this gap in the literature by focusing on the role of religious principles and Muslim religious (Shari'ah) scholars, along with education, social identity, culture, socioeconomic status, demographics and the influence these factors have on Somalis regarding the need and motivation for using student loans while allowing their attitude towards these loans to be further explored. Examining the role of religious principles and the influence of Shari'ah scholars in this context is essential because student loans are interest-bearing financial instruments and Somalis consider a religious prohibition on interest-bearing financial instruments as a part of their religious identity (Change Institute, 2009; Christensen et al., 2004; Mews & Abraham, 2007). This religiosity-based group identity takes on economic, social, cultural and symbolic forms and consequently affects the developmental trajectory of Somali communities in England (Muhamad, 2008; Vertovec, 2004). Their disinclination towards interest-based student loans owing to religious values is largely acknowledged by the British Government (BIS, 2014) and within academic literature (Abdul-Muhmin, 2008; Aida & Imen, 2014; Vogel & Hayes, 2006). However, the extent of its influence in causing exclusion from higher education has not been systematically explored.

There are between 2.7 and 3.0 million Muslims in the UK, with approximately 230,000 Muslim students attending universities—around 8%–9% of the total student population (Guest et al., 2020). Nearly half of Muslims (48%) were aged under 25 (1.3 million), an increase of 505,000 since 2001 (Galadima, 2015). Historically, the social formation of Islamic society did not necessitate non-productive loans during the days of the Prophet; it had not been reflected upon to prohibit a charge on commercial and productive loans (EI-Gamal, 2003). The majority of scholars believe that any increase in monetary value (whether small or large) stipulated in a loan contract, irrespective of whether it is a consumption loan or a productive loan, is *ribā* and so is prohibited by the Shari'ah (Siddiqi, 2004). Some scholars (Ibn Jibreen, 2006) have argued that a student may proceed to accept a student loan when no other alternative is available on the basis of 'need' or 'necessity'. However, there has been much controversy and confusion among the Islamic scholars and followers of Islam, based on the definition of 'need' or 'necessity' (Al-Qudah, 2009; Az-Zair Salahudin, 2010).

Overall, interest-based student loans have had a significant impact on thousands of Muslim students in the UK. Stevenson (2014) shows that the introduction of the interest-based student loan policy has stopped thousands of potential Muslim students from going to university. In a consultation run by BIS (2014), from a sample of 20,000, '94% of respondents stated that there would be a demand for an alternative finance product which was Sharia-compliant'. Morris (2016) also found that the UK has a significant demand for Islamic finance education funding. Through the *Higher Education and Research Act 2017*, the *Takaful programme*

1469351 (8, Downloaded from https://otm-journals.olimelibrary.wiley.com/doi/10102bej3347 by NHS Education for Southand NES, Edubtrary for level Commons License

4 | **BERJ**

for Islamically compatible financing of HE should have commenced in the academic year 2016/17; however, it is currently delayed until 2025 (UK Parliament, 2022).

Overall, the government policy documents do not acknowledge that, within the current system, people belonging to the same community and living in the same location will not have the same level of access to educational opportunities because of intersecting inequalities based on their religion, gender, socioeconomic status and (dis)abilities (Deka, 2004; Hanson, 2006). Our research asks whether religious principles on financial transactions and the positions of Shari'ah scholars on student loans are creating barriers to the education of the Somali community. From the perspective of rational economic behaviour (Etzioni, 1987), one can describe the Somali Muslim community as using the rule of thumb in making decisions (Bertrand et al., 2005) on whether to engage with student loans, because this decision becomes a moral judgement based on religious values. Our research also examines the impact and influence of the two separate theological approaches to the positive framing of student loans by the Shari'ah scholars (groups A1 and group B—Table 1).

Currently, there are no studies that attempt to explore the influence of religion and the magnitude of preference for Islamic finance-styled instruments on the use of student loans for higher education. There is also a lack of narratives on the broader topics of equality, inclusivity and widening access to higher education in the context of this conflict between minority communities' religious values and the financial structure of student loans (Cabinet Office, 2007). There are currently no analytical enquiries on the influencing factors that may underpin the decision-making process of the Somali Muslim community on the use of student loans, such as the role of Shari'ah scholars, social norms (Bundy & Pfarrer, 2015; Calder, 2019; Carr, 2005; Davies & Lea, 1995; Vossensteyn & De, 2006), peer pressure (Lamborn et al., 1996; Mladenovic et al., 2017), student debt (Eaton et al., 2016), finan-

	Position on interest	Position on student loan	Institutions/Shari'ah scholars
Group A1	Interest-bearing products are prohibited	<i>Permissible:</i> student loans are permissible owing to the doctrine of necessity	 The European Council for <i>Fatwa</i> and Research – Ireland and England Assembly of Muslim Jurists of America – USA Islamic <i>Fiqh</i> Academy – India S. Ibn Jibreen – KSA Dr Muzammil Siddiqi – USA
Group A2	Interest-bearing products are prohibited	<i>Prohibited</i> : Student loans do not meet the criteria of doctrine of necessity	 Australian Institute of Islamic Jurisprudence (Darul Ifta Australia) – Australia Permanent Committee for Scholarly Research and <i>Ifta</i>' of the Council of Senior Scholars – KSA <i>Darul Ulum</i> Karachi – Pakistan S. Salih Asuheymee – KSA
Group B	Interest-bearing products are different from classic usury and are permissible	<i>Permissible</i> : Student loan does not include usurious transaction	 Azhar Islamic Research Institute – Egypt S. Mohamed Sayed Tantawi – Grand Mufti of Egypt Professor Fazle Rehman – University of Karachi, Pakistan Mohamoud El-Gemal – Rice University, USA

TABLE 1	Taxonomic cataloguing of religious opinions (fatawa) on student loans

Source: Authors' own collection of fatawa from official institutions and personal websites of scholars.

cial literacy (Lee & Mueller, 2014) and other cultural influencers (Bapuji & Chrispal, 2018). A difference of opinion also exists among Shari'ah scholars on the religious prohibition of student loans (El-Gamal, 2003). For instance, a good proportion of local and international religious scholars allow the use of interest-bearing student loans for higher education—which suggests that socially constructed rejection of student loans may not be solely based on a religious—ethical judgement. Some studies, such as Ismail (2009), Ismail et al. (2014) and the Secretary of State for Business, Innovation and Skills (2015), propose Islamic finance-styled student loans. However, the potential effectiveness of this proposal in addressing the issue remains unknown, owing to the lack of bottom-up discussions and consultations with the Somali community. Contemporary literature (academic and government reports) lacks co-constructed narratives on the perceptions held by Somali Muslim communities regarding the issues surrounding the use of student loans as an effective means of accessing higher education.

This study proposes exploring the influence of factors such as religious values, preference for Islamic financial instruments, verdicts of Shari'ah scholars (*fatawa*), socioeconomic profile, education and demographics on the use of student loans within the Somali Muslim community and their resulting behavioural outcomes, which may allow us to fully understand the barriers to higher education faced by the Somali community in England.

To assess the issues outlined above, this article is organised as follows: Section 2 briefly discusses the profile of Somalis living in England, and Section 3 then examines ethical values and religious verdicts (*fatawa*) on the use of student loans. Section 4 presents the details on data collection and the methodological approach, while Section 5 discusses the research findings by focussing on the impact of historical, religious and ethical values, along with *fatawa* on accepting or rejecting the student loans by Muslim Somalis from five major cities in England. Section 6 contextualises the use of student loans and situates the research findings within the broader literature, and proposes a theoretical underpinning to elucidate Muslim Somali behaviour, which extends beyond neoclassical rationality. The final section presents the concluding remarks.

THE SOMALI COMMUNITY IN ENGLAND

Somalis first arrived in England during the late nineteenth century and settled around the naval-port cities (Anya, 2014). A second wave of Somali migrants arrived during the Second World War (Kahin, 1997) because of the high labour demand for the steel industry in Sheffield and South Yorkshire (Hammond, 2013a). The Somali civil war in the 1980s further displaced Somalis, who eventually settled in the UK after fleeing refugee camps in Ethiopia, Djibouti, Kenya and Yemen (Hammond, 2013a, 2013b). The majority of the families that have arrived in the UK in the 2000s have transnational relatives still living as refugees, and Government efforts to rebuild the Somali community in the UK have further attracted Somalis displaced in other parts of Europe (Moret, 2016).

Somali communities settled in the UK have living memories of war and fleeing homelands owing to conflict, which was then followed by a cycle of dependencies, interdependencies and independencies owing to multiple displacements (Special Report, BBC, 2020) and then (re)settlement into the radically different and sometimes hostile socioeconomic culture of England (Kushner, 2006). These living memories of past experiences have dramatically shaped the current economic and cultural situation of Somali communities living in England.

Currently, Somalis living in England are categorised as one of an underrepresented and deprived community discussed within the widening accessibility and financial inclusivity literature (Stevenson & Baker, 2018). England currently hosts the largest Somali population outside of Somalia (Harris, 2004; Holman & Holman, 2003), with settlements predominantly 6 | BERJ

in London, Birmingham, Manchester, Sheffield and Leicester (Hassan et al., 2013). Somalis living in England have consistently high rates of unemployment, low income and low educational attainment (Demie et al., 2007, 2019; Pollard et al., 2019; Rasmussen, 2011; Valentine et al., 2009).

Somali communities in England face serious problems of constant 'exclusion' from higher education (Change Institute, 2009: 54; Pollard et al., 2019: 2). The full scale of the problem becomes apparent when we compare the literacy rates. The UK has a literacy rate of 99%, which means that one in every 100 people struggles to read and write; however, the literacy rate of Somalis living in the UK is approximately 41% (Bloch & Atfield, 2002; Pollard et al., 2019), which means that 60 in every 100 people struggle to read and write. The majority of the 40 in every 100 Somalis that do achieve secondary education do not continue to tertiary higher education, resulting in one of the lowest participating ethnic groups in higher education within the UK (Hammond, 2013a).

The literature lists the reasons for low participation in higher education as the increase in tuition fees, a lack of Scottish-style state support (Stevenson, 2014) and high living costs (Middlemiss, 2017), thus making it financially impossible for Somalis to fund their higher education (BIS, 2012). However, these factors may apply to all ethnic minorities, while they also essentialise the situation of Somalis living in England. In particular, Somalis living in England have the lowest employment rates and education levels of any migrant group in the UK (Open Society Initiative for Europe; At Home in Europe & Open Society Institute, 2015). Over 50% hold no qualifications and only 3% attain higher education qualifications (Kyambi, 2005).

HYPOTHESES DEVELOPMENT: STUDENT LOANS, SHARI'AH SCHOLARS AND POSITIVE FRAMING

About 89.3% of Somalis in England are Sunni Muslims (Change Institute, 2009) and they live in a close-knit community (Allport et al., 2019) which functions within dynamic social, linguistic and cultural structures that are grounded within the classical religious traditions of Islam. The charging of interest is a key financial component of modern-day student loans and it is prohibited on religious grounds within the Islamic tradition. This religious ban on usury (*ribā*) is a key pillar of the Islamic system of morality (Choudhury, 1997; Mews & Abraham, 2007). The religious understanding of moral judgements on usury (*ribā*) is embedded within the Islamic culture and is part of the generalised morality that is generated by the community (Ismail, 2010). Over centuries, this understanding has manifested itself within the social moral norms and has become a part of the social structure, culture, identity and notion of self-integrity. This understanding exists as social knowledge (Turiel, 1983) and is independent of Shari'ah scholars and Islamic religious institutions; therefore, we are referring to it as a cultural value.

Since 1970, Shari'ah scholars have attempted to positively frame the historic prohibition of interest. Some of them (group B in Table 1) frame the modern interest rate as permissible, because they argue that it is different in principle from the historical practice of usury (*ribā*) that is prohibited within Islam (EI-Gamal, 2003; Rahman, 2009). Others (group A1 in Table 1) frame student loans as permissible based on the theological principle of necessity (2010; Ismail, 2010; AI-Qudah, 2009; Ibn Jibreen, 2006, Islamic Fiqh Academy in India, 2010). The positive framing by group A1 (Table 1) is based on the overarching theological principle of the protecting human public interest, welfare, well-being and social wealth (AI-Qudah, 2009; Opwis, 2007; Safian, 2010; Vogel & Hayes, 1998). There are also groups of scholars (group A2 in Table 1) who align themselves with the historic stance and argue for a blanket religious prohibition on the use of interest within financial instruments (AI-Mahmudi, 2017;

As-Suheymee, 2011; Usmani, 2002). Table 1 taxonomically groups these Shari'ah scholars on their stance on using student loans for higher education.

The moral internalisation of these religious verdict-based positions on the use of usury (*ribā*) is strengthened through the process of religious socialisation by the associated religious group led by each Shari'ah scholar within the transnational Somali communities (Robinson, 2010). While the literature and government reports on the use of student loans only consider the stance of group A2, this study integrates the positionings of groups A1 and B, along with A2, to explore their effects on the decision-making process of Muslim Somalis.

The religious rulings by these three groups of Shari'ah scholars are grounded within the four constructs of Islamic legal ethics, that is necessity of student loans (*dharura*), interest on student loans (*ribā*), objective of student loans (*maqāsid*) and the broader benefit of student loans to the community (*maslaha*). The Shari'ah scholars use these constructs to justify their ethical positioning on the religious acceptability or rejection of student loans. These four constructs are also grounded as moral foundations within religious values (Al-Qudah, 2009; El-Gamal, 2003; Ibn Jibreen, 2006; Ismail, 2009; Rahman, 2009, 2010).

It is relevant to mention here that the religious positioning and discrepancies with regard to student loans is informed by the Shari'ah scholars' positioning and repositioning based on different determinants of *fatawa* (Ullah et al., 2016). This is because there is no universal or generally accepted understanding of how to transform the descriptive Shari'ah rules into a system of checkable investment guidelines (Derigs & Marzban, 2008).

The conflict between the host society and the minority immigrant community's religious values (Mukhametzaripov & Kozlov, 2017) seems to be a critical emerging issue. However, the literature (Allen et al., 2014; Ball et al., 2002) on accessibility, inclusivity and widening access to higher education points to other barriers that may cause exclusion and inequality for the Somali Muslim communities' access to higher education, including: anxiety over long-term debt (Sawyer & Carroll, 2000), financial need for immediate income streams (Furlong & Biggart, 1999), influence of living in the least affluent neighbourhoods (Knowles, 1999) and lack of support from family and parents who mostly have not received higher education (Metcalf, 1997).

Therefore, the study hypotheses are:

- **Hypothesis 1** The ethical premise on the morality of student loans is set within society by the religious values of Somali Muslim communities living in England.
- **Hypothesis 2** The participants' perception of the purpose of student loans is affected by their level of adherence to the religious values.
- **Hypothesis 3** The Shari'ah scholars and their verdicts influence Somali Muslim communities' perception of the benefit of using student loans to fund higher education for Somali youth.
- **Hypothesis 4** The participants perceive that the moral positions of the ethical values that prevail within the Somali Muslim communities living in England reflect the moral positions of the verdicts of Shari'ah scholars.
- **Hypothesis 5** The participants perceive the verdicts of Shari'ah scholars as reflective of the moral positions of the religious–ethical values that prevail within the Somali Muslim communities living in England.
- **Hypothesis 6** Shari'ah scholars and their verdicts influence the attitude of Somali Muslim communities towards the use of student loans.

To operationalise the research, a number of latent variables are used from the existing literature related to the demand for Islamic financing in various cases. For example, Demirguc-Kunt et al. (2017) find little use of Shari'ah-compliant banking products, although there is evidence of a hypothetical preference for Shari'ah-compliant products among a plurality of respondents despite the higher costs. Similarly, the study of the demand for halal mortgages in the UK explores the role of religion to determine the consumer preferences for alternative mortgages (Tameme & Asutay, 2012) or loans (Borchgrevink & Erdal, 2022).

EMPIRICAL MODEL SELECTION AND ESTIMATION METHOD

Data

This exploratory study is a community-based participatory research enquiry that takes an interpretivist approach based upon a social constructivist paradigm. The purpose of this jointly constructed study is to explore the factors, in particular the role of Shari'ah scholars and community religious values, in Somali Muslims' (1) attitude towards student loans and (2) perception of the purpose of using/rejecting student loans and whether religious values are influenced by Shari'ah scholars.

The exploration undertaken in this study applies co-construction, a collaborative approach to data collection and knowledge generation. In this process of co-construction and participation, the researchers from Edinburgh Napier University, Al-Maktoum College of Higher Education and Abertay University took a 'step back cognitively' approach from all means of data collection, forms of interaction and power relationships (Bergold & Thomas, 2012). To address power relationships, all interactions and data collection were conducted through a Somali *imam* (who leads Muslim worshippers in prayer). He is a Somali community leader and minister of religion with no social affiliations to groups A1, A2 and B Shari'ah scholars and holds a neutral personal stance on the subject of religiosity and the usage of student loans. An informal discussion on the use of student loans for higher education was held with the Somali community leader and representatives from Somali communities in London, Birmingham, Manchester, Sheffield and Leicester. Interestingly, the four concepts of Islamic legal norms [necessity of student loans (*dharura*), interest on student loans (*ribā*), the objective of student loans (maqāsid) and the broader benefit of student loans (maslaha) to the community] emerged as key themes within the content of these discussions, along with the individual Shari'ah scholar's verdict as a socioethical force that influences the perceptions of Somali Muslims. Therefore, the impact of Shari'ah scholars' fatawa, along with these four constructs were used in questionnaire design to assess the bearing on Somalis' decision-making regarding student loans. In total, the questionnaire is based on six panels (constructs; latent variables)-panel A, attitude towards student loans; panel B, religious values; panel C, perception on the purpose of student loans; panel D, Shariah scholars; panel E, communal solutions based on communal ethos; and panel F, preference for Islamic finance-styled student loans—and 29 observed variables (indicators) spread across the six constructs. The questions use a Likert scale, which is anchored from 1 (strongly agree) to 5 (strongly disagree).

The neutrality and anonymity of participants of the survey was ensured by collecting all questionnaires through a physical drop-box placed at the community centres. The informal initial discussions with the community were facilitated but not mediated by the Somali *imam*. Prior to the discussion, the Somali *imam* received training on neutrality, data collection, data protection and data management. This study made use of questionnaires instead of interviews, because of a clear preference favoured by the Somalis consulted during the initial discussions with the *imam* during the participatory design process. The *imam* distributed 550 questionnaires out of which 190 were returned and used for analysis by the research team. The participants sampled for primary data are derived from diverse socioeconomic backgrounds. Data collection was conducted, through co-exploration with Somali communities, between May 2018 and July 2019 in the five English cities. The geographic areas sampled

BERJ | 9

were selected according to the demographic weight of Somali populations living in English cities. Table 2 provides a statistical distribution of Somalis in five English cities: London, Birmingham, Manchester, Sheffield and Leicester. The table shows that more than half of the Somali community live in London.

Table 3 presents the profile of the Somali respondents who participated in the study. It shows that more than half of the sample population (55.3%) are active university students, while 23.2% are not studying and 1.1% are not planning to attend higher education at all, while a further 16.8% are planning to study. This indicates that 24.3% of the sample may not be attending higher education, with a further 16.8% who are coming of age also not planning to attend higher education.

With respect to academic achievement, the table shows that the population represents an almost even distribution of participants within the different educational levels, with 13.7% (GCSE and A level) aspiring to enter higher education, 64.7% already in some form of university education and 21.6% involved in some other forms of education. Finally, regarding socioeconomic status and place of birth, there appears to be a broad range of participants within the cohort. Participants were asked to self-identify their socioeconomic status based on income, educational attainment, financial security, self-perception of social status and quality of life (Kincheloe & Steinberg, 2007). Over half (67.4%) of the participants self-identified as

City	Number of Somali populations	Percentage of Somalis in Muslim population of city	Percentage of Somalis in population of area	Number of questionnaires distributed	Number of returned questioners
London	131,000	12%	1.51%	200	76
Birmingham	63,000	10%	4.85%	100	52
Manchester	19,000	6%	0.68%	75	17
Leicester	11,000	9%	2.63%	100	26
Sheffield	9000	6%	1.24%	75	19

TABLE 2 Distribution of Somali population in England

Source: extracted from UK Populations, https://ukpopulations.com and Furqan Mosque Leicester Database for the Somali community.

T/	٩B	L	Е	3	Profiles of respondents
----	----	---	---	---	-------------------------

University st	atus	Gender		Age rai	nge	Academic achievem	-	Socioeconomio status	•	Place of birth	
	Ν		N		Ν		Ν		Ν		Ν
Currently student	105	Female	91	Below 18	16	GCSE	6	Working/lower class	124	Born in the UK	34
Not student	44	Male	99	18–20	48	A level	20	Upper working/ lower class	4	Born outside the UK and EU	13
Planning to study	32			21–24	65	College	16	Middle class	56	Born in the EU	131
Not planning to study	2			25–30	31	Bachelor	87	Upper middle class	5	Born in Somalia	12
None	7			31–40	26	Master	20	High class	1		
				Above 40	4	Other	41				

Source: Authors' questionnaire survey.

working class, which is the socioeconomic group where the cost of tuition fees and maintenance living cost act as an economic barrier to entry into higher education.

Study models and estimation methods

Our proposed model, the UK Somali Muslims Acceptance of Interest-bearing Student Loans, is presented by a path model (Figure 1a,b), which represents the relationship between the study constructs and their observable indicators (Bollen, 2002). The panel-A in the Figure 1 explains the impact of religious values on the rest of the variables, while panel-B illustrates the impact of Shari'ah scholars on the other variables.

The path model was estimated using partial lease squares structural equation modelling (PLS-SEM) owing to the lack of theoretical developments within the field of our enquiry (Hair et al., 2012). Since the primary focus of the research is to explore the factors influencing the attitude and perception of Somali Muslim communities regarding the use of student loans in accessing higher education, the PLS-SEM was deemed to be the most preferable modelling choice (Hair et al., 2017; Rigdon, 2012). The PLS-SEM allows the questions (observed indicators) in the six panels to be anticipated as a latent construct (Hair et al., 2011; Hair Jr et al., 2014), distributed by first order in Appendix 1. The PLS-SEM provides complex cause–effect structural models (Richter et al., 2016; Rigdon, 2012, 2014) with many indicators suitable for analytical methods that are required to measure the impact of religious values (panel B) and Shari'ah scholars' verdicts and positions (panel D). The PLS-SEM also allows for soft modelling with the flexibility to accommodate distributional assumptions for studies with data that are not normally distributed owing to the sample size (Hair et al., 2017; Wold, 1980).

The path model compromises two models: the reflective measurement model (inner), which captures the relationship between the observed indicators and their corresponding latent variables (constructs); and the structure model (outer), which captures the relationship between the latent variables (constructs) and hence reflects the hypothesised relationships (Sarstedt et al., 2017).

The reflective measurement model is illustrated by the following equation (Sarstedt et al., 2017):

$$X = \beta^* L \tag{1}$$

where X is an indicator (any of the 29 indicators in the study), L is the corresponding latent variable (any of the six latent variables) and β is the regression coefficient that represents the relationship between the observed indicator and the latent variable.

In order to evaluate the results of the PLS-SEM, first the reflective measurement model (measurement theory) is evaluated using: (1) indicator reliability (loadings 0.70 or above are considered reliable; Hulland, 1999); (2) internal consistency reliability of the constructs measured by (a) Joreskog's (1971) composite reliability (above 0.6 is considered reliable) and (b) Cronbach's alpha convergent validity measured by average variance extracted across all the constructs' indicators (0.5 or higher is acceptable); and (3) discriminant validity, measured by Henseler et al.'s (2016) heterotrait–monotrait ratio of correlations (Sarstedt et al., 2014; Sarstedt et al., 2017).

The structural model which entails computation of the path coefficients (between constructs) is based on several regression analyses.

$$Y_i = \lambda^* Y_j \tag{2}$$





12 | BERJ

where Y_i is the attitude towards student loans, perception of the purpose of student loans or preference for Islamic finance-style student loans, while Y_j is either religious values or Shari'ah scholars' *fatawa*.

If the quality of the reflective measurement model is supported, the structural model (structural theory), which includes hypothesis testing, is evaluated by: (1) testing collinearity among latent variables using VIF (above 5 indicates collinearity); (2) evaluating the coefficient of determination (R^2) which measures how much of the variance of the endogenous latent variable is explained, and anything the change in R^2 (f^2 effect size of path coefficient); (3) the predictive relevance Q^2 (higher than 0 is acceptable) and any changes in Q^2 (q^2 effect size of path coefficients) and; (4) the relevance and significance of path coefficients (hypotheses testing) including the total effects (Sarstedt et al., 2014, 2017).

RESULTS

Factors shaping the Somali' behaviours: Descriptive statistics

The fundamental argument of group A scholars is acceptability of student loans based on financial needs (panel A of Table 4). The survey confirms that 49% of Somalis are using or have used student loans for education, whereas 41% neither use nor plan to use student loans. Most Somalis (84%) also do not consider student loans to be a helpful financial tool for inclusion into higher education. Consequently, as many as 64% would not seek the support from the student loan companies to achieve their educational goals. The Somali community is recognised for its generosity in financial support from relatives and friends, and 47.4% of the sample expect such support from their relatives. Such social practices may reduce the financial need for student loans (Pollard et al., 2015). Some 50.6% of Somalis favour outright rejection of the use of student loan arrangements, while 9.5% of participants remain undecided.

Panel B captures the participants' understanding of Islamic religious norms, as they shape Somali behaviour in relation to the student loans. Some 93.4% of the participants showed a good understanding of the religious prohibition of usury (*ribā*); however, only 60% of the sample participants perceive modern day interest-bearing financial instruments as unethical on religious grounds. This finding is in agreement with Siddiqui (2005). The discrepancy between these two percentages shows the impact of the religious opinions of groups A1 and group B in opposition to the conservative stance of group A2 scholars. Even so, the data suggest that group A2 scholars have far more outreach and influence on the Somali community, as 55% of participants align with group A2, while only 9% oppose them. Interestingly, 35% are neutral and undecided—we will explore further the dynamics of this influence within the next section.

Only 35% of the survey population have confidence in the enacted approach of ecclesial representatives on student loans of groups A1 and B (panel D of Table 4), whereas 80.5% are unconvinced. Some 29.5% of the survey population accept interest-based loans and align with the concept of necessity of group A1, whereas 70% disagree or remain neutral. Equally, 75.8% are unable to justify the use of student loans based on a social benefit, which is an argument of group A1. As many as 50.5% of the survey population would consider avoiding a student-based loan to safeguard their religion, thus aligning with group A2, yet it is interesting to note that 41.6% remain uncertain.

Some 76.8% of participants disagree on the responsibility of the Somali community to provide financial support to Somali youth for access to higher education (panel E of Table 4). Equally 70.5% of the survey population do not feel it is the responsibility of the scholars to provide alternative solutions to the student loans for gaining access to higher education.

BERJ | 13

TABLE 4 Survey data

		centage who ongly agree			age who ⁄ disagree	
<i>N</i> = 190	1	2	3	4	5	Mean
Panel A: Attitude towards Student Ioan						
I have used/been using student loan arrangements	36	13	10	8	33	2.88
For my study official financial loans are very important	5	5	5	29	55	4.26
For studying, you will receive financial support from relatives and friends	17	18	12	23	29	3.28
Financial schemes from student loan companies are very important	9	9	17	27	38	3.76
Panel B: Religious values						
You are familiar with <i>ribā</i> (usury) term	62	32	3	1	3	1.50
Interest-based loans are prohibited like usury	41	20	31	7	2	2.10
It is Islamically prohibited to take interest-based loan hence student loan	30	25	35	7	2	2.25
In student loan contract there is breach of Islamic principles	5	8	35	25	27	3.61
All conditions of student loan beyond interest are compliant with Islamic law	7	22	60	7	4	2.78
Panel C: Perception on Purpose of Student Loan						
Taking the student loan is primarily for paying fees	26	42	22	5	5	2.20
Taking the student loan is primarily for life maintenance	9	29	19	25	18	3.14
Taking a loan was essential to complete my education	38	29	11	13	10	2.26
In case of not taking a student loan your education will not be fulfilled	22	29	23	18	8	2.62
Taking a student loan contributed to your financial inclusion	14	30	33	12	12	2.77
Panel D: Influence of Shari'ah scholars						
I am confident in the permissibility of student loans based on available <i>fatawa</i>	20	15	47	14	4	2.68
I am not confident in the permissibility of student loans based on available <i>fatawa</i>	4	9	46	26	16	3.41
Permissibility of student loans is justifiable on the basis of <i>darura</i> (necessity)	13	17	41	24	6	2.92
Permissibility of student loans is justifiable on the basis of <i>maslaha</i> (social benefit)	12	20	44	20	4	2.83
Avoiding student loan schemes is pre-requisite for safeguarding the religion	1	7	42	31	21	3.63
The permissibility of student loan based on <i>darurarh</i> (necessity) is rejected	1	10	61	18	11	3.27
Panel E: Communal solutions based on communal ethos						
Providing the alternative to student loan is duty of Muslim community	2	7	18	37	41	4.11
Shari'ah scholars are responsible to suggest such community-based solution for a student loan	2	7	20	37	34	3.94
Providing Shari'ah-compliant student loan is essential for Somali student	2	2	11	32	53	4.11

(Continues)

14 | BERJ

TABLE 4 (Continued)

		centage who ongly agree			age who ⁄ disagree	
<i>N</i> = 190	1	2	3	4	5	Mean
Zakah funds should be used for student needs in England not abroad	41	37	18	3	2	1.88
Panel F@ Preference for Islamic finance style student loans						
There is no difference between Islamic or conventional student loan	1	5	14	27	54	4.27
Having Shari'ah scholars approve a student loan gives you confidence	41	28	18	11	2	2.04
Shari'ah-compliant student loan more expensive is preferable to conventional	40	31	20	6	3	2.01
Based on my experience with student loan I will encourage others to use it	5	17	25	23	30	3.56
It is Somalin community duty to establish charity for investing in our youth	52	35	10	2	2	1.30

Source: Authors' Questionnaire survey.

Some 85.3% of the survey population agree that an Islamic version of a student loan is not an essential element when questioned specifically about an alternative source such as *Zakah* (compulsory alms), while 40% agree that it would be beneficial and 33.7% do not agree that *Zakah* sources should be used for student loan purposes. Some 67.3% of survey participants would accept the ruling of Shari'ah scholars on the matter of loans and 70% of the survey population agree that even if a more expensive alternative to a conventional interest-based loan was available, they would not use it. As many as 54.8% of the participants would not advise others to use interest-based loans, creating a considerable peer effect in dealing with interest-based loans. The survey finally suggests that 85.3% of Somalis would prefer to have an alternative channel for financing higher education in the form of an Islamic charity dedicated to this purpose.

Age, gender and social conformity pressure: Inferential statistics

This section cross-references the control categories of university status, gender, age range, academic achievement, socioeconomic status, and place of birth with the construct indicators. The Kruskal Wallis and Mann–Whitney U tests were used to analyse the dataset for possible statistically significant differences among the groups (*See* Table 5).

Our data show that, like all societies, there exists a conflict in the permanence of differences within the generations of the Somali Muslim community. There is a difference in interpretation and perception on the necessity of student loans across generational lines; the age-effect shows a *p*-value of 0.048. The participants aged 31–40 (with a mean ranking of 75.25), accept the permissibility of student loans based on necessity, thus aligning with group A1 scholars. However, they also perceive (mean ranking 112.37) that modern-day interest is not prohibited—hence simultaneously aligning with group B scholars. The participants in the age groups 18–20 and 21–24 (mean rankings 107.82 and 105.61, respectively) strongly acknowledge the importance of student loans for paying fees as well as the maintenance living costs with a *p*-value of 0.006. Furthermore, the participants within the group aged 18–30 with high mean rankings (94.13, 111.44 and 92.17) express a lack of confidence

Motion fractional matrix fractinal matrix fractional matrix fractional matrix fractional matrix		University	University status					Gender			Age range	ge					
Current base Bead		Mean ran	king					Mean rai	nking		Mean ra	inking					
indet(autr)(rm) 646 032 617 032 647 032 647 032 647 032 647 032 647 032 647 032 647 032 043 032 0436 <th></th> <th>Current students</th> <th>Non- students</th> <th>Planning on studying</th> <th></th> <th></th> <th>p-value</th> <th>Female</th> <th>Male</th> <th>p-value</th> <th>B 18</th> <th>18–20</th> <th>21-24</th> <th>25-30</th> <th>3140</th> <th>A 40</th> <th>p-value</th>		Current students	Non- students	Planning on studying			p -value	Female	Male	p -value	B 18	18–20	21-24	25-30	3140	A 40	p -value
and on some prohubic 71 0.50 0.81 <td>ou are familiar with <i>ribā</i> (usury) term</td> <td>94.64</td> <td>103.22</td> <td>85.81</td> <td>131</td> <td>80.43</td> <td>0.347</td> <td>102.97</td> <td>84.54</td> <td>0.006</td> <td>92.97</td> <td>90.45</td> <td>90.52</td> <td>104.56</td> <td>105.65</td> <td>86.75</td> <td>0.571</td>	ou are familiar with <i>ribā</i> (usury) term	94.64	103.22	85.81	131	80.43	0.347	102.97	84.54	0.006	92.97	90.45	90.52	104.56	105.65	86.75	0.571
intentione primerior 10-46 52-5 53-55 11-7 22-5 53-55 12-1 22-5 53-55	terest-based loans are prohibited e usury	87.12	105.58	108.03	98.75	85	0.162	98.74	88.99	0.193	102.4	86.94	82.92	107.45	112.37	151	0.013
involutionationationationationationationationa	king the student loan is primarily r paying fees	100.46	85.22	85.95	84.75	119.71		86.09	102.34	0.03	94.57	107.82	105.2	80.08	66.38	78.63	0.006
is satisfiely of student lange2.36100.2017.5017.	strmissibility of student loans is stifiable on the basis of <i>darura</i> ecessity)	100.9	85.35	83.38	41	121.79		89.45	97.82	0.267	93.1	91.25	107.61	94.97	75.25		0.048
10342 76.17 69.29 97.76 107 97.3 91.69 92.36 69.88 97.76 97.75 91.75 91.69 92.37 60.84 97.75 91.75 91.75 91.75 91.75 91.75 91.75 91.75 91.75 91.69 92.17 91.75 9	te permissibility of student loan sed on <i>darurarh</i> (necessity) is jected	82.36	108.28	110.03	77.5	109	0.005	100.52	84.89	0.024	102.4	88.62	83.19	90.95	123.48		0.004
10:00: 5:34 9:15 10.75 <th< td=""><td>oviding Shari'ah-compliant student an is essential for Somali students</td><td>103.42</td><td>76.77</td><td>85.92</td><td>97.75</td><td>109.21</td><td></td><td>87.84</td><td>99.53</td><td>0.101</td><td>86.17</td><td>96.63</td><td>98.69</td><td>92.35</td><td>86.88</td><td>97.75</td><td>0.889</td></th<>	oviding Shari'ah-compliant student an is essential for Somali students	103.42	76.77	85.92	97.75	109.21		87.84	99.53	0.101	86.17	96.63	98.69	92.35	86.88	97.75	0.889
10:00 81.78 73.00 10 121.71 0.018 92.96 90.56 90.56 96.47 85.52 11 0 Actoments Actoments Actoments 90.56 90.56 90.56 90.56 90.57 85.52 11 0 0 9	σ	105.02	75.94	82.22	97.5	107.5	0.017	89.32	97.96	0.252	94.7	94.13	111.44	92.17	63.73	40.5	0.001
Address and the set of the		101.08	81.78	78.06	150	121.71		92.96	93.04	0.991	103.53	96.86	90.56	96.47	85.52		0.771
Mean rankingdear rankingMean rankingdear rankingMean rankingdear rankingMean rankingdear rankingmean rankingmean rankingdear rankingmean rankingmean rankingmean rankingdear rankingmean rankingmean rankingmean rankingmean rankingmean rankingmean rankingdear rankingmean rankingmean rankingmean rankingmean rankingmean rankingmean ranking<		Academ	ic achieven	lent				Soc	io-econom	ic status			E	ace of birth	_		
GCSF A College BA Master Other Pavalue WL WL WL WL MS Pavalue Somalia Somalia 104 97.43 84.41 91.4 93.24 105.35 0.718 94.53 63.13 97.36 MG MC MC<		Mean ra	nking					Меа	in ranking			 	Ň	aan ranking			
104 97.43 84.41 91.4 93.24 105.35 0.718 94.53 63.13 97.36 13.45 95.65 93.55 96.25 17 105.13 78.03 96.22 98.21 95.31 0.537 102.8 154.8 95.65 93.55 96.25 7.4.17 105.13 78.03 96.22 98.21 95.35 102.83 68 81 71.4 56 0.029 113.45 85.54 94.68 70.5 70.42 84.9 103.13 96.29 94.07 95.89 0.773 103.75 75.5 76.25 715.4 50.5 105.8 95.65 93.33 85.33 100.03 92.59 98.85 103.85 86.13 79.88 64.9 40 0.024 110.58 92.69 93.26 93.24 87.54 87.54 87.54		GCSE	٨	College						MC	UMC				Somalis & EU		p -value
74.17 105.13 78.03 96.22 98.21 93.21 0.537 102.83 68 81 71.4 56 0.029 113.45 85.54 94.68 70.5 70.42 84.9 103.13 96.29 94.07 95.89 0.773 103 75.5 76.25 115.4 50.5 0.016 110.58 92.6 93.33 85.33 100.03 92.59 98.85 103.85 86.13 79.88 64.9 40 0.024 119.18 92.12 94.24 87.67	ave used/been using student loan arrangements	104	97.43	84.41							100.8					96.25	0.05
70.42 84.9 103.13 96.29 94.07 95.89 0.773 103 75.5 76.25 115.4 50.5 0.016 121.28 110.58 92.6 93.33 85.33 100.03 92.59 98.85 103.85 86.13 79.88 64.9 40 0.024 119.18 92.12 94.24 87.67	or my study official financial loans are very important		105.13							81	71.4					70.5	0.235
85.33 100.03 92.59 98.85 104.1 85.75 0.343 103.85 86.13 79.88 64.9 40 0.024 119.18 92.12 94.24 87.67	rancial schemes from student loan companies are very important	70.42	84.9							76.25	115.4	5				93.33	0.028
	king the student loan is primarily for paying fees	85.33	100.03	92.59							64.9					87.67	0.178

(Continued)
ß
ш
_
Ω
◄
\vdash

	Academ	Academic achievement	ent					Socio-ec	Socio-economic status	status				Place of birth	f birth			
	Mean ranking	nking						Mean ranking	ıking					Mean ranking	nking			
	GCSE	٩	College	BA	Master	Other	p-value	MLC	UWL	MC	UMC	거	p -value	Я	EU	Outside UK, Somalia & EU	Somalia p -value	p -value
I am confident in permissibility of student loans based on available fatawa	73	86.68	120.53	95.29	99.5	87.03	0.237	89.14	48.5	106.91	120.2	108.5	0.055	94.85	78.73	98.98	86.07	0.271
Permissibility of student loans is justifiable on the basis of <i>maslaha</i> (social benefit)	72	88.23	97.41	91.99	105.93	99.54	0.69	94.55	49.5	101.61	36	162.5	0.015	91.58	93.5	94.39	87.94	0.982
Providing Shari'ah-compliant student Ioan is essential for Somali students	68.83	103.18	96.75	100.4	105.03	75.18	0.058	97.27	117.88	87.17	97.7	57.5	0.534	87.65	87.27	97.61	85.36	0.539
Having Shari'ah scholars approve a student loan gives you confidence	48.83	111.9	83.44	100.45	117.9	72.58	0.001	101.34	83.5	82.07	77.1	87.5	0.192	95.7	58.5	101.51	67.06	0.013
Shari'ah-compliant student loans that are more expensive are preferable to conventional	98.17	88.98	81.5	101.7	89.28	86.89	0.545	97.75	115.63	84.74	101	37	0.342	108	71.81	96.45	57.28	0.044
N= 190.																		

Note: The significance test is based on 95% or higher significance levels i.e. a p-value of 0.05. All insignificant questions were deleted.

1

in the Shari'ah approval of existing student loans (*p*-value 0.01), whereas the participants aged 31+ accept Islamic scholars' verdicts as a confidential tool for decision-making regarding student loans.

The strong acknowledgement of the importance of student loans suggests an aspiration and predilection to pursue higher education within the Somali Muslim community. However, the overall low presence of Somalis within higher education (Jones et al., 2010) and the existence of a conflict on the morality of using student loans across generational lines suggests the existence of a conformity effect (Bond & Smith, 1996), given that 'moral decision-making is strongly influenced by social context' (Kundu & Cummins, 2013: 276). The portrayal of religiosity through conformity and compliance with social norms is an essential cultural element of the close-knit Somali Muslim community, because the conformity provides inclusion in social capital networks of the community (Mulcahy et al., 2010). However, the overall role of religious morality, social inclusion and conformality on the behaviours and attitudes is a complex and contingent issue that cannot be reduced to straightforward quantification (Wakefield & Poland, 2005).

Gender, age and academic achievements, when cross-referenced with other questions, show statistically significant differences—thus 'The permissibility of student loans based on *dharurarh* (necessity)' is rejected. Females show a greater inclination towards rejecting student loans based on the notion of need with an overall p-value of 0.024 and a mean ranking 100.52. However, religious knowledge on interest prohibition (ribā) (p-value 0.006) in female participants is lower than that in male participants. Females are also more likely to accept student loans only for paying fees (with a mean ranking of 86.09), which is not the case with male participants (mean ranking 102.34). Gender differences in behaviour and educational preferences (among others) have been documented in the literature. Croson and Gneezy (2009) report that women are less risk tolerant, have more variability in social behaviour and show more opposition to competition than men. It has also been found that women generally speak slightly slower than men (Jacewicz et al., 2009) and demonstrate better reading competencies than men, which has resulted in men having a lower enrolment in tertiary education compared with women (Stoet & Geary, 2020). Interestingly, however, men have displayed stronger relationships between religious activities and well-being than women (Maselko & Kubzansky, 2006).

Generally, there is more social pressure to conform and comply for Somali women than for men (Shaffer, 2018). The difference in gender conformity may be related to the psychosocial and cultural context embedded in the transnational sociocultural, economic and religious norms (Scuzzarello & Carlson, 2019; Toder & Marcia, 1973; Weinschenk et al., 2018). A large body of empirical research, such as Berns et al. (2005), Deuker et al. (2013) and Mallinson and Hatemi (2018), underpinned by Asch (1951, 1955) and Sherif et al. (1961), shows that social conformity pressure has a significant impact on behaviours and attitudes and it diminishes the sensitivity to monetary benefits (Yu & Sun, 2013). The intensity of social conformity pressure on women may be creating a need for women to show higher religiosity by taking a conservative stance on student loans—the higher religiosity factor in women owing to social conformity pressures and the gender imbalances are consistent with other studies, such as Walter and Davie (1998), Chamberlain and Zika (1988), Diehl et al. (2009), Ghaffari and Çiftçi (2010) and Glick et al. (2016), among others.

The case for an underlying theme of social conformity pressure is further supported by the significance (*p*-value of 0.044) of preferences towards Islamic law-compliant student loans that are slightly more expensive compared with their conventional counterparts. Participants born in the UK show strong disinclination (mean ranking 108) towards paying more for an ethically tailored financial instrument, while participants born in Somalia show a strong preference (mean ranking 57.28) for such a setup. This aligns significantly (*p*-value of 0.013) with 'having Shari'ah scholars approve a student loan gives you confidence', where UK-born

participants showed strong disagreement (mean ranking 95.7) in comparison with a general agreement among Somali-born participants (mean ranking of 67.06). This could be contextualised within intrinsic and extrinsic religiousness; our data align with the work of Vitell et al. (2005) on the dynamics of religiosity and consumer behaviour. The difference in preferences between the UK-born Somali participants and others could be because the UK-born Somali may be viewing the ethicalness of 'consumer practices ... [based on] how one's religion is viewed by others' (Vitell et al., 2005: 179). This could be due to the pressure of social conformity to adhere to the socially established ethical judgements regarding the financial instruments in question. The adherence to social norms is essential for accessing the social capital and social support from the community (Beverland et al., 2010; Bizri, 2017).

Participants born in the UK may not require access to social capital that could leverage an influence on socially established norms, so therefore the characteristics that formulate their ethical–religious judgement are different. However, ethical–religious judgement on the consumption of student loans for the participants born in Somalia is grounded within the religious opinions of Shari'ah scholars (mean ranking of 67.06) and they are also willing to pay more for an ethically acceptable financial instrument (mean ranking 57.28). Moreover, the Somali-born participants (mean ranking 93.33) consider the financial arrangements of student loans (*p*-value 0.028) to be more important for inclusion into higher education than the UK-born participants (mean ranking 121.28).

The socioeconomic status is also significant (p-value of 0.029) with regard to interacting with student loans; unexpectedly the use of conventional financial support for higher education is important for those of high class (HC) (mean ranking 56), in contrast to those from the working lower class (WLC; mean ranking 102.83), who surprisingly find it of least importance. The WLC, who arguably have the most need for student loans in order to be able to access higher education, consider adherence to religious values far more important than access to higher education. Both the middle class (MC) and HC are open to religious acceptance of interest-based student loans, (mean ranking of 101.61 and 162.5, respectively) and least aligned with the historic stance on religious prohibition of all interest-bearing financial instruments (Souaiaia, 2014). In contrast the upper middle class (UMC; mean ranking 36) and upper working class (UWC; mean ranking 49.5) are open to accepting the verdict of modern Shari'ah scholars (group B) that theologically justifies the interest-bearing student loan on the grounds of social benefits (El Gemal, 2003). However, there is a clear class divide among the social classes (p-value 0.055), where the WLC (mean value of 89.4) and UWC (mean value of 48.5) show confidence in current religious verdicts on student loans, while the MC (mean value of 106.91), UMC (mean value of 120.2) and HC (mean value of 108.5) show strong disagreement. This could be due to higher conformity pressure on WLC and UWC, as compared with the upper classes. There is also a significant (p-value = 0.024) inversely proportional relationship between social classes and their perception on the purpose of student loans; for instance, the HC (mean ranking 40) only sees paying tuition fees as a purpose of student loans, while as we go down the social class scale the perception exponentially changes and at the WLC (mean ranking 103.85) there is a strong perception that tuition fees should cover expenses beyond tuition fees, such as maintenance and subsistence. This coincides with the comprehension of broader financial burdens on participants below the MC category.

The UWL (mean ranking 68.0) and HC (mean ranking 56.0) strongly agree on the importance of student loans to pursue higher education (*p*-value 0.029), in addition to strongly suggesting (mean ranking 75.5; mean ranking 50.5) the importance of financial schemas by financial institutions (*p*-value 0.016). However, the WLC (mean ranking 102.83) strongly deemphasises the importance of student loans to pursue higher education (*p*-value 0.029), in addition to strongly downplaying (mean ranking 103.0) the importance of financial schemas by financial institutions (*p*-value 0.016). Interestingly, the suggestion of religious alms

BERJ

(*Zakah*), along with the financial support from family and friends for higher education, was insignificant across the board. The significant deemphasis on the importance of student loans within the WLC aligns with the findings of other studies, including Connor et al. (2001), Archer et al. (2005), Maloutas (2007) and Sianou-Kyrgiou and Tsiplakides (2011) among others.

The research report of Connor et al. (2001) also discovered a high level of dissatisfaction within current student financial support instruments. The study by Furlong and Biggart (1999) shows that the WLC has relatively less financial literacy and is generally uninformed about the financial instruments used to provide student loans. The lack of importance shown for these financial instruments within our sample by WLC group could also be due to the lack of understanding of these instruments (Arthur, 2012).

The academic achievement also has an impact on the perception towards student loans, as the participants currently at GCSE level (mean ranking 68.83) show strong agreement on student loans that are religiously compliant (*p*-value 0.058). In contrast strong disagreement with the essentiality of religiously compliant student loans for access to higher education decreases in participants with higher degrees. This may be due to Somali students with more experience within education realising the complex barriers, beyond financial needs, to entry into higher education (Allen et al., 2014; Ball et al., 2002).

The current students (mean ranking 101.08) strongly disagree with paying extra for Shari'ah-compliant student loans (*p*-value 0.018), while the participants that are planning on studying (mean ranking 78.06) and non-student participants (mean ranking 81.78) show strong inclination towards paying more for a Shari'ah-compliant student loan. Participants who are non-students (mean ranking 81.78) and participants planning on studying (mean ranking 81.78) and participants planning on studying (mean ranking 81.78) tend to agree on the necessity of Shari'ah-compliant student loans (*p*-value 0.03) and reliance on Shari'ah scholars' verdicts on the permissibility of student loans (*p*-value 0.017), while current student participants from our sample (mean ranking 103.42 and 105.02) tend to strongly disagree. The discrepancy in perception between students and non-students aligns with the differences in opinions based on the age group, further suggesting the complexity of the issue of student loans and inclusion into higher education.

Results of estimating the UK Somali Muslims' acceptance of interest-bearing student loans—path models

This part of the empirical analysis presents the results of estimating the UK Somali Muslims' path models and the impact of religious values and influence of Shari'ah scholars, using the PLS–SEM method. The results are presented in Figure 2a, b and Tables 6 and 7, respectively. The first step is to assess the reflective measurement (outer) model reliability and validity (Hair Jr et al., 2014) by evaluating indicator reliability, internal consistent reliability, convergent validity measures and discriminant validity.

The observable variables (indicators) measuring a reflective construct must be unidimensional in order to be considered unique values (Gefen et al., 2000). As shown in Table 6, unidimensionality is confirmed by retaining variables whose loadings are above 0.4 (Hulland, 1999). Indicators with loadings between 0.4 and 0.7 are retained because they are statistically significant, even though their contribution to explaining the construct is relatively low. Furthermore, each of the construct's average variance extracted is above 0.5 (Table 7), which supports its convergent validity (Fornell & Larcker, 1981; Gefen et al., 2000; Hair Jr et al., 2014; Henseler, et al., 2016). All observable variables that did not meet the indicator loading thresholds were removed.

Internal consistency reliability is more appropriately measured by composite reliability rather than Cronbach's α , because the former does not rely on the assumption of equal load-



20

BERJ



FIGURE 2 UK Somali Muslims' acceptance of interest-bearing student loans: (a) impact of religious values; and (b) UK Somali Muslims' acceptance of interest-bearing student loans-impact of Shari'ah scholars. Partial least squares structural equation modelling. SmartPLS automatically composed these charts. The outer weights/ loadings and structural path coefficients are shown with arrows and R² in round factors (detailed results are shown in Table 6).

TABLE 6	Measurements of model-convergent validity (after removing the unsignificant panels and
dimensions)	

	Religious v	values		Shari'ah sc	holars	
	Loading	t-Statistics	<i>p</i> -Values	Loading	t-Statistics	<i>p</i> -Values
a1 ← Attitude	0.972	5.230	0.000	-0.872	1.509	0.132
a2 ← Attitude				0.609	1.713	0.087
a3 ← Attitude	-0.501	2.491	0.013			
a4 ← Attitude	-0.580	3.035	0.003	0.802	1.642	0.101
b1 ← Religious values	0.439	4.696	0.000			
b2 ← Religious values	0.909	52.470	0.000	0.921	76.579	0.000
b3← Religious values	0.899	43.840	0.000	0.924	69.633	0.000
$c1 \leftarrow Perception$	0.555	3.528	0.000	0.557	3.542	0.000
$c3 \leftarrow Perception$	0.928	19.079	0.000			
$c4 \leftarrow Perception$	0.627	3.753	0.000	0.603	3.020	0.003
$c5 \leftarrow Perception$				0.855	5.960	0.000
d1 \leftarrow Shari'ah scholars	0.746	2.795	0.005	0.828	26.626	0.000
$d2 \! \leftarrow Shari'ah \ scholars$				-0.735	11.009	0.000
$d3 \! \leftarrow \! Shari'ah \ scholars$	0.829	2.814	0.005	0.836	30.330	0.000
$d4 \! \leftarrow \! Shari'ah scholars$	0.742	2.846	0.005	0.721	12.569	0.000
d5 ← Shari'ah scholars	-0.776	2.634	0.009	-0.723	15.465	0.000
d6← Shari'ah scholars	-0.780	2.616	0.009			

TABLE 7 Construct's reliability and validity

	rho_A	Average variance extracted
Panel A	Religious values	
Attitude	1.640	0.511
Religious values	0.792	0.609
Perception	0.895	0.521
Shari'ah scholars	0.838	0.601
Panel B	Shari'ah Scholars	
Attitude	0.815	0.592
Religious values	0.826	0.851
Perception	0.203	0.547
Shari'ah scholars	0.833	0.593

ings for the indicators and does not underestimate internal consistency in a way the latter does (Hair Jr et al., 2014). However, the composite reliability tends to overestimate the actual reliability; therefore, Dijkstra and Henseler (2015) have proposed (ρ_A) to overcome the deficiency of composite reliability, with a threshold of 0.7 (Sarstedt et al., 2017), hence, values of ρ_A are calculated and reported in Table 7. In both models, religious values (panel B) and Shari'ah scholars (panel D), the majority of the constructs' ρ_A exceeded 0.7. Furthermore, the discriminant validity measured by the heterotrait–monotrait ratio of correlations proposed by Henseler et al. (2016) meets the threshold, which is less than 0.85 (Sarstedt et al., 2017), as presented in Table 8.

Having established the reliability of the reflective measurement (outer) models, now we move to evaluate the results of the structural (inner) models, i.e. hypothesis testing. Figure 1a depicts the UK Somali Muslims' acceptance of interest-bearing student loans with a focus on the impact of religious values, while Figure 1b depicts the same model but examines the impact of Shari'ah scholars' verdicts and positions. In addition, Table 9 presents the path coefficients and their *t*-statistics, which represent the relationships between the constructs of the study. As shown in Figure 1a and Table 9, the religious values of Somali Muslim communities have an impact on their attitudes towards interest-bearing loans. This negative relationship suggests that the religious–ethical values embedded within the Somali culture inform and dictate the participants' attitudes and view on the importance of student loans and decisions to use them. This supports H1 and is consistent with the findings of other studies, such as BIS (2014), Christensen et al. (2004) and the Secretary of State for Business, Innovation and Skills (2015). However, these studies do not separate the religious–ethical values embedded within the culture from the religious verdicts by Shari'ah scholars on the use of student loans.

Furthermore, from Table 6 it can also be seen that indicators b_2 and b_3 have strong relationships with their religious value construct. This demonstrates that participants' religious values are in agreement with the historical–conservative blanket prohibition on the use of any interest-bearing financial instrument, and rejection of the theological justification provided by group A1.

The result as shown in Figure 1a, demonstrates that perception of the purpose of student loans has a negative but weak relationship with religious values, supporting H2. This indicates that participants are aware of the purpose of student loans and their importance, and that religious values have a significant impact on this perception. The strong relationship of indicator c_3 'taking a loan was essential to complete my education' with its construct and perception on the purpose of student loans, shows that the participants had a good understanding of the essentiality and need for student loans for higher education. This suggests

	Religious values	Shari'ah scholars
Perception → attitude	0.987	1.416
Religious values → attitude	0.372	0.266
Religious values \rightarrow perception	0.327	0.312
Shari'ah Scholars →attitude	0.520	0.427
Shari'ah Scholars \rightarrow perception	0.489	0.777
Shari'ah Scholars \rightarrow religious values	0.779	0.694

TABLE 8 Discriminant validity – heterotrait–monotrait ratio

TABLE 9 The relationships between the constructs – hypothesis testing

Hypothesis	Path coefficients	t-Statistics	Hypothesized relationship	Conclusion
1. Religious values \rightarrow Attitude	-0.333	3.778***	Negative	supported
2. Religious values \rightarrow Perception	-0.265	4.900***	Negative	Supported
3. Shari'ah scholars \rightarrow perception	-0.312	4.430***	Negative	Supported
4. Shari'ah scholars \rightarrow Religious values	0.579	10.015***	Positive	Supported
5. Religious values \rightarrow Shari'ah scholars	0.605	2.691***	Positive	Supported
6. Shari'ah scholars \rightarrow attitude	0.375	1.476	Negative	Not supported

Note: p<0.10; *<0.05; ***p<0.001.

that the participants are aware of the impact of not using student loans on the entrance to higher education. However, in spite of this understanding, the participants are driven by the religious-ethical values, as these values are shaping their decision-making process.

Therefore, the low participation of Somalis in higher education (Jones et al., 2010) is not due to a lack of understanding on the use of financial instruments. This is contrary to the findings of Furlong and Biggart (1999), Archer (2003), Van Zanten (2017), Connor et al. (2001), Maloutas (2007), Sianou-Kyrgiou and Tsiplakides (2011), Archer et al. (2005) and others that low financial literacy, lack of understanding of student loans and general de-emphasis on higher education within the lower income group are contributing factors to the lower participation in higher education. Our data suggest that comprehension of the significance of using student loans for entering higher education is not the determining factor for Somali community engagement with student loans. This could be because the decision making on engagement with student loans seems to be built on moral judgement, which is an outcome of conscious deliberation based on religious norms. The religious–moral principles constructed on the 'social consensus' seem to guide the decision-making behaviour of the Somali community (Kant, 1785, 1787; Klucharev et al., 2009; Kohlberg, 1969; Kundu & Cummins, 2013).

The Shari'ah scholars construct captures the framing effect (De Martino et al., 2006) as the Shari'ah scholars in group A1 and B are re-framing the historical prohibition on interest-bearing instruments (student loans), while group A2 aligns itself with religious values embedded within the Somali culture. The results (Table 9) show the positive effect of religious values on Shari'ah scholars supporting H5, and Table 6 shows that d_1 , d_3 , and d_4 all have nearly the same power, whereas d_5 and d_6 have a negative relationship with the construct. This suggests that the religious–moral context of interest-bearing student loans by Shari'ah scholars. The strong influence of religious values held in high priority by the Somali Muslim community suggests that the participants align themselves with the verdicts of Shari'ah scholars that are coherent with their religious values. The ineffectiveness of the (re)framing of student loans is consistent with the social identity effect, as explored in the works of many including (but not limited to), Hogg (1996, 2016), Sierra and McQuitty (2007), Kramer and Brewer (1984) and Brewer and Kramer (1986).

Figure 1b and Table 9 show that the participants' perception of the benefits of using student loans is unfavourably impacted by Shari'ah scholars and their verdicts, supporting H3. Furthermore, the results also strongly support H4 and indicate that the participants who identify with religious values perceive that the social morality of the Somali Muslim community reflects the verdicts of religious scholars, which is consistent with the findings of El-Gamal (2003).

Finally, H6 is not supported, which suggests that despite the manner in which participants formulate their religious–ethical identity, their attitude towards student loans is not influenced by the verdicts of religious scholars. This shows that the moral norms of religious value embedded within the social structure, culture, identity and notion of self-integrity are constructing the moral judgement on the ethicality of student loans (Hogg, 1996; Turiel, 1983). These moral judgements are not strongly affected by the verdicts of religious scholars.

INFLUENCE OF SOCIAL ORIGINS AND THE POWER OF ECONOMIC RATIONALITY

The enquiries into the process of decision making grounded in the neo-classical approach are underpinned within the framework of the rational, utility-oriented and profit-maximising homo-economicus, while the developments within behavioural geography attempt to humanise this homo-economicus (Hanson, 2006; Strauss, 2008). Our study sits within the discourse that attempts to humanise the narrative on decision making. Our data suggest that to explain the behaviour of participants one should go beyond the narrative of economic rationality. Our data show that the decision-making of the Somali Muslim community on the use of student loans for higher education is intersected with their identity and spatiality, and the social and cultural embeddedness of their religious values. This is consistent with the work of Barnes and Sheppard (1992), Ettlinger (2004), McDowell (2005), Miller (1992), Peet (2000), Smith et al. (2006), Strauss (2008) and Sunley (2000) among others.

From a social psychology perspective, the behaviour of the Somali Muslim community may be categorised as what Lewis (2017) describes as belief preservation bias. Our data shows that the Somali Muslim community has a reasonably good understanding regarding the importance of student loans for the potential role they have in enabling the attainment of higher educational objectives/aspirations. Our findings are consistent with the existing research on the relationship between beliefs and reasoning biases (Klaczynski, 2000; Klaczynski & Fauth, 1997; Klaczynski & Gordon, 1996; Klaczynski & Lavallee, 2005). This study suggests that in making their decisions on the use of student loans, the Somali Muslim community tend not to take into account either their perception of the significance of student loans in advancing their careers or the theologically positive framing by Shari'ah scholars, suggesting that any further positive framing by the Government, higher education institutions or Shari'ah scholars to attract the Somali Muslim community within the current student loan model may not work.

This is consistent with the experience of Islamic banks in the UK; these banks were established to provide financial inclusivity to Muslims living in the UK. Islamic banks provide financial instruments that are financially engineered and theologically framed (Calder, 2019; Laldin & Furqani, 2015; Usmani, 2002), similar to the approach of group A1 and group B1 scholars (Table 1). Islamic banks also incorporate Shari'ah boards within their internal governance structures to support the positive framing of their financial instruments and to signal the religiosity of their operations. However, most Muslims living in the UK do not use these banks or instruments, as they discount the positively framed theological narrative about the permissibility of these instruments, similar to the participants in our study (Karbhari et al., 2004).

The financial engineering and the positive reframing based on theological reasoning have been critiqued by both academics and practitioners (Asutay, 2012; El-Gamal, 2009; Khaleel & Avdukić, 2020). Our data suggest that any financial instruments which re-configure interest-bearing student loans through theological framing and financial engineering as interest-free financial instruments (similar to contracts used in Islamic banks) may also not achieve the intended purpose.

Despite the ongoing discussions to improve access to higher education for BAME students, the policy implications within the British context have not been fully explored (Cleland et al., 2018; Searle et al., 2018). The decision-making of the Somali Muslim community, in the context of student loans, is a complex process that is influenced by variabilities of spatial, cultural, religious and economic conditions. Our data suggest that Somali Muslim communities living in England may not have as much free agency or access to benefit from higher education opportunities as some other ethnic communities residing in the same locations. Our research has also shown that the Somali Muslim community remains under the structural influence of social conformity pressures to adhere to religious values. These dynamics are contributing to the underlying factors responsible for the exclusion of Somalis from higher education, further causing the social inequality and marginality of the Somali Muslim community in England. The social conformity pressure plays a key role within the dynamics of this disenfranchising structure. Therefore, any reduction of the social conformity pressure, in this context, may improve the access of the Somali Muslim community to higher education.

Taking into account the issue of 'conformity is as old as humanity' (Sunstein, 2019: IX), there have been long-standing calls for exploring ways to reduce the social conformity

pressures that allow the recalibration of social norms, from different academic disciplines (Binning et al., 2015; Prentice, 2008; Reid et al., 2015; Reid et al., 2019). Individuals give in to social conformity pressure to corroborate social affiliation and validate the accuracy of their decision-making (Chartrand & Lakin, 2013; Cialdini & Goldstein, 2004; Cialdini & Trost, 1998; Reid et al., 2019). In the context of the Somali Muslim community, they seek validation of their decision-making within the social norms governed by religious values. The authoritative cultural norms have a stronger influence on individuals' behaviour, decisions and actions (Binning et al., 2015; Petty & Cacioppo, 1986). Our data (Table 5) suggest that there is an implicit relationship between socioeconomic vulnerability, social class and adherence to social conformity pressure. Reflection on the results suggests that the economic behaviour of the Somali Muslim community, in the context of student loans, should not be rationalised only on the factor of religiosity of the community, since social pressures and social validity of the decision-making seem to play a greater role than mere religiosity in dictating behaviour.

The implications of this study for the larger (multi-ethnic) Muslim community are that future research enquiries and government policy documents that rationalise the economic behaviour of these communities should not focus solely on religiosity as the decisive factor. It is clear that micro-cultures of the community and its socially constructed ethical worldview(s) have implications for economic decision-making. There is also a degree of generalisability of this study to all the ethnic minority groups and communities living in the UK. The majority of ethnic minority groups in the UK practise a sub-culture that is different from the dominant culture. Therefore, government policies that focus on improving the participation of these ethnic minorities should first consider rationalising the perceptions of the community, along with contextualising their decision-making, within the sub-culture of that community. Any solutions or policies made thereafter should also be constructed in conjunction with that ethnic minority.

This study indicates that the Somali Muslim awareness and knowledge of Islamic principles play a crucial role in forming the perception of student loans offered by the student loan agencies. This study also confirms the findings of previous studies that religious norms are important factors for Somali Muslims to accept a Shari'ah-compliant student loan. However, as our satisfaction analysis indicates, this is not a sufficient condition for their decision-making regarding Shari'ah alternatives.

Hence, our analysis suggests that there is no simple solution (for example Islamic finance-styled student loans) that could address this issue. Islamic banks have attempted to co-create financial solutions with Shari'ah scholars, but our data suggest that positive reframing by Shari'ah scholars may not work. To create a solution, in the context of student loans, interaction with the Somali Muslim communities may provide a more fruitful endeavour. Key stakeholders should provide a participatory role (Cova & Dalli, 2009; Ind et al., 2012) to Somali Muslim communities for the co-construction of a customised solution (Franke & Piller, 2004; Syam & Pazgal, 2013). Such a solution should respond to their socioeconomic and religious needs to create a holistic approach for widening their access to higher education.

CONCLUSION

This study co-explores the view of Somali Muslims on their perception of need, motivation and attitude towards student loans, including their preference of Islamic financial instruments, and the impact of religious values and Shari'ah scholars on their decision-making process (see Tables 4 and 5). Using a combination of descriptive statistics, inferential statistics and PLS-SEM, community-based participatory research enquiries were performed on Somali Muslim community data from five English cities and subsequently analysed.

Our data sample shows that the offering of Islamic financial instruments for student loans will not resolve the exclusion of Muslim minorities from higher education. This is based on finding that the ethical positioning of Islamic financial principles does not align with the ethical preference of the new generation of Muslim Somalis.

We also co-examined the relationship between the impact of ethical religious values from panel B (Table 4), with the influence of Shari'ah scholars in panel D (Table 4) and the perception of Somalis on the use of student loans for inclusion into higher education. The findings demonstrate that financial understanding of the need, purpose and use of student loans, along with the verdicts of Shari'ah scholars are not the determining factor. Instead, the social origin and the ethical religious values of the community (panel B) (Table 4) act as the determining factor for the refusal to use student loans, which creates the underlying issue that causes exclusion from higher education.

The results of this study offer several implications. The findings show the atheoretical nature of discourse present on the (dis)engagement of Somali Muslims with student loans for funding higher education. For Shari'ah scholars, this research shows that the religious verdicts somewhat influence the community developments, although there is less susceptibility to the effect of positive reframing and general rejection of Islamic finance-styled financial solutions. The results also highlight the need for policy-makers and higher education institutions to engage with the Somali Muslim community to co-find solutions for inclusion and widening access to higher education. This is essential, because our data reject the general understanding that Muslim minority communities showcase a demand for Shari'ah-compliant financial instruments and their absence causes the financial exclusion of these communities. Our results show that ethical religious values can act as cultural and social barriers to development, which results in people sharing the same location being unable to share the same level of opportunities because of their religious beliefs.

FUNDING INFORMATION

The authors declare that they have not obtained any funding for conducting the research.

CONFLICT OF INTEREST

The authors declare that they have no conflict of interest.

DATA AVAILABILITY STATEMENT

The data are available upon reasonable request.

ETHICS STATEMENT

All procedures performed in studies involving human participants were in accordance with the ethical standards of the institutional and/or national research committee and with the 1964 Helsinki Declaration and its later amendments or comparable ethical standards. Informed consent was obtained from all individual participants involved in the study.

REFERENCES

Abdul-Muhmin, A. G. (2008). Consumer attitudes towards debt in an islamic country: Managing a conflict between religious tradition and modernity? *International Journal of Consumer Studies*, *32*(3), 194–203.

- Aida, K., & Imen, K. (2014). Islamic finance for future entrepreneurs: Exploring Tunisian business students knowledge and potential usage of Islamic financing products. *Journal of Emerging Economies and Islamic Research*, 2(3), 43–56.
- Allen, F., Carletti, E., Carletti, E., Carletti, E., Cull, R., Qian, J. Q. J., Qian, J. Q. J., & Valenzuela, P. (2014). The African financial development and financial inclusion gaps. *Journal of African Economies*, 23(5), 614–642.

BERJ | 27

- Allport, T., Mace, J., Farah, F., Yusuf, F., Mahdjoubi, L., & Redwood, S. (2019). 'Like a life in a cage': Understanding child play and social interaction in Somali refugee families in the UK. *Health and Place*, 56, 191–201.
- Al-Mahmudi, M. (2017). Are student loans a Mudharabah (investment contract). Retrieved from http://darulfiqh.com/ are-student-loans-a-mudharabah-investment-contract. Accessed November 2, 2019.
- Al-Qudah, D. M. K. (2009). In Reality of and ruling on student loans in The United States of America (p. 29). Assembly of Muslim Jurists in America. Retrieved from http://www.amjaonline.org/en/declarations/20-declarations/52-decisions-and-recommendations-of-amja-s-sixth-annual-convention-canada. Accessed October 1, 2019.
- Anya, A. (2014). 100 years after Somalis arrive in London, it's high time to learn about them. Voices, Open Society Foundation Retrieved from https://www.opensocietyfoundations.org/voices/100-years-after-somalis-arrivelondon-it-s-high-time-learn-about-them. Accessed August 3, 2020.
- Archer, L. (2003). Social class and higher education. Higher education and social class: Issues of exclusion and inclusion. Routledge.
- Archer, L., Hutchings, M., & Ross, A. (2005). *Higher education and social class: Issues of exclusion and inclusion*. Routledge.
- Arthur, C. (2012). Financial literacy education (pp. 1–12). Brill Sense.
- Asch, S. E. (1951). Effects of group pressure upon the modification and distortion of judgments. In H. Guetzkow (Ed.), *Groups, leadership and men; research in human relations*. Carnegie Press.
- Asch, S. E. (1955). Opinions and social pressure. Scientific American, 193(5), 31-35.
- As-Suheymee, S. S. (2011). Fatwa on student loans. Retrieved from https://miftaahulilm.wordpress.com/2011/11/29/ fatwa-on-student-loans-shaykh-saalih-as-suhaymee/. Accessed November 4, 2020.
- Asutay, M. (2012). Conceptualising and locating the social failure of Islamic finance: Aspirations of Islamic moral economy vs the realities of Islamic finance. *Kyoto Bulletin*, *9*, 93–113.
- Az-Zair Salahudin, W. (2010). Dawābid al-Hājah allatī tanzilu manzilati darūrah wa tadbīqātihā alal ijtihādāt al-Mu'asirah. (Principles of Evidence on Revelation: Orders of Necessity, Their Applications in Construction of New Rulings for Contemporary Times). Dar Al-Birr.
- Bachan, R. (2014). Students' expectations of debt in UK higher education. *Studies in Higher Education*, 39(5), 848–873.
- Ball, S. J., Reay, D., & David, M. (2002). Ethnic Choosing': Minority ethnic students, social class and higher education choice. Race Ethnicity and Education, 5(4), 333–357.
- Bapuji, H., & Chrispal, S. (2018). Understanding economic inequality through the lens of caste. *Journal of Business Ethics*, *162*(3), 533–551.
- Barnes, T. J., & Sheppard, E. (1992). Is there a place for the rational actor? A geographical critique of the rational choice paradigm. *Economic Geography*, 68, 1–21.
- BBC Special Report. (2020). Born Abroad—Somalia. Retrieved from http://news.bbc.co.uk/1/shared/spl/hi/uk/05/ born_abroad/countries/html/somalia.stm. Accessed May 1, 2020.
- Bergold, J., & Thomas, S. (2012). Participatory research methods: A methodological approach in motion. *Historical Social Research*, 37(4), 191–222.
- Berns, G. S., appelow, J., Zink, C. F., Pagnonni, G., Martin-Skurski, M. E., & Richards, J. (2005). Neurobiological correlates of social conformity and Independence during mental rotation. *Biological Psychiatry*, 58(3), 245–353.
- Bertrand, M., Karlan, D. Mullainathan, S. Shafir, E., & Zinman, J. (2005). "What's psychology worth? A field experiment in the consumer market", Center Discussion Paper, No 918, Yale University, Economic Growth Center, New Haven, CT.
- Beverland, M. B., Farrelly, F., & Quester, P. G. (2010). Authentic subcultural membership: Antecedents and consequences of authenticating acts and authoritative performances. *Psychology and Marketing*, 27(7), 698–716.
- Binning, K. R., Brick, C., Cohen, G. L., & Sherman, D. K. (2015). Going along versus getting it right: The role of self-integrity in political conformity. *Journal of Experimental Social Psychology*, 56, 73–88.
- BIS (2012). Education (Student Loans) (Repayment) (Amendment) (No. 2) Regulations 2012: Equality Impact Assessment, (2). Retrieved from https://www.gov.uk/government/publications/education-student-loans-repayment-amendment-no-2-regulations-2012-equality-impact-assessment. Accessed October 29, 2019.
- BIS. (2014). Government response to consultation on a sharia- compliant alternative finance product. Webb.
- Bizri, R. M. (2017). Refugee-entrepreneurship: a social capital perspective. *Entrepreneurship and Regional Devel-opment*, 29, 847–868.
- Bloch, A., & Atfield, G. (2002). The professional capacity of nationals from the Somali regions in Britain. Refugee Action & IOM.
- Bollen, K. A. (2002). Latent variables in psychology and the social sciences. *Annual Review of Psychology*, 53(1), 605–634.
- Bond, R., & Smith, P. B. (1996). Culture and conformity: A meta-analysis of studies using Asch's (1952b, 1956) line judgment task. *Psychological Bulletin*, 119(1), 111–137.

28 | BERJ

- Borchgrevink, K., & Erdal, M. B. (2022). The necessity of taking a loan, and the necessity of not doing so: Islamic finance among Muslim minorities in Denmark and Norway. *Nordic Journal of Religion and Society*, *34*(2), 117–129.
- Brewer, M. B., & Kramer, R. M. (1986). Choice behavior in social dilemmas: Effects of social identity, group size, and decision framing. *Journal of Personality and Social Psychology*, 50(3), 543–549.
- Bundy, J., & Pfarrer, M. D. (2015). A burden of responsibility: The role of social approval at the onset of a crisis. Academy of Management Review, 40(3), 345–369.
- Cabinet Office. (2007). Fairness and freedom: The final report of the equalities review. Crown.
- Calder, R. (2019). How religio-economic projects succeed and fail: The field dynamics of Islamic finance in the Arab gulf states and Pakistan, 1975–2018. *Socio-Economic Review*, *17*(1), 167–193.
- Carr, N. (2005). Poverty, debt, and conspicuous consumption: University students tourism experiences. *Tourism Management*, 26(5), 797–806.
- Chamberlain, K., & Zika, S. (1988). Religiosity, life meaning and wellbeing: Some relationships in a sample of women. *Journal for the Scientific Study of Religion*, 27(3), 411–420.
- Change Institute. (2009). The Somali Muslim Community in England: Understanding Muslim ethnic communities. Communities and Local Government, 2009. http://www.communities.gov.uk/documents/communities/ doc/1202162.doc Accessed November 18, 2019.
- Chartrand, T. L., & Lakin, J. L. (2013). The antecedents and consequences of human behavioral mimicry. *Annual Review of Psychology*, *64*, 285–308.
- Choudhury, M. A. (1997). Money in Islam: A study in Islamic political economy. Routledge.
- Christensen, P. N., Rothgerber, H., Wood, W., & Matz, D. C. (2004). Social norms and identity relevance: A motivational approach to normative behavior. *Personality and Social Psychology Bulletin*, 30(10), 1295–1309.
- Cialdini, R. B., & Goldstein, N. J. (2004). Social influence: Compliance and conformity. *Annual Review of Psychology*, 55, 591–621.
- Cialdini, R. B., & Trost, M. R. (1998). Social influence: Social norms, conformity and compliance. In D. Gilbert, S. Fiske, & G. Lindzey (Eds.), *The handbook of social psychology* (Vol. 2, 4th ed.). McGraw-Hill.
- Cleland, J. A., Patterson, F., & Hanson, M. D. (2018). Thinking of selection and widening access as complex and wicked problems. *Medical Education*, *52*(12), 1228–1239.
- Connor, H., Dewson, S., Tyers, C., Eccles, J., Regan, J., & Aston, J. (2001). Social class and higher education: Issues affecting decisions on participation by lower social class groups. Great Britain.
- Cova, B., & Dalli, D. (2009). Working consumers: The next step in marketing theory? *Marketing Theory*, *9*(3), 315–339.
- Croson, R., & Gneezy, U. (2009). Gender differences in preferences. Journal of Economic Literature, 47(2), 448–474.
- Czarnecki, K., Korpi, T., & Nelson, K. (2020). Student support and tuition fee systems in comparative perspective. *Studies in Higher Education*, 46, 2152–2166. https://doi.org/10.1080/03075079.2020.1716316
- Davies, E., & Lea, S. E. G. (1995). Student attitudes to student debt. *Journal of Economic Psychology*, *16*(4), 663–679.
- De Martino, B., Kumaran, D., Seymour, B., & Dolan, R. J. (2006). Frames, biases, and rational decision-making in the human brain. *Science*, *313*(5787), 684–687.
- Deka, D. (2004). Social and environmental justice issues in urban transportation. In S. Hanson & G. Guiliano (Eds.), The geography of urban transportation. Guilford Press.
- Demie, F., Lewis, K., & McLean, C. (2007). *Raising the achievement of Somali pupils*. Lambeth Research and Statistics Unit.
- Demie, F., Taplin, A., Butler, R., Tong, R., McDonald, J., & Hau, A. (2019). *The achievement of Somali pupils in Lambeth schools—Empirical evidence 2018*. Lambeth Research and Statistics Unit.
- Demirguc-Kunt, A., Klapper, L. F., & Randall, D. (2017). Islamic finance and financial inclusion: Measuring use of and demand for formal financial services among Muslim adults. In *World Bank Policy Research Working Paper*, *No. 6642*, World Bank. Available at: https://openknowledge.worldbank.org/bitstream/handle/10986/16875/ WPS6642.pdf?sequence=1&isAllowed=y
- Derigs, U., & Marzban, S. (2008). Review and analysis of current shariah-compliant equity screening practices. International Journal of Islamic and Middle Eastern Finance and Management, 1(4), 285–303.
- Deuker, L., Muller, A. R., Montag, C., Markett, S., Reuter, M., & Fell, J. (2013). Playing Nice: A multi-methodological study on the effects of social conformity on memory. *Fronteirs in Human Neuroscience*, 7(79), 1–11.
- Diehl, C., Koenig, M., & Ruckdeschel, K. (2009). Religiosity and gender equality: Comparing natives and Muslim migrants in Germany. *Ethnic and Racial Studies*, 32(2), 278–301.
- Dijkstra, T. K., & Henseler, J. (2015). Consistent partial least squares path modeling. *Mis Quarterly = Management Information Systems Quarterly*, 39(2), 297–316.
- Eaton, C., Habinek, J., Goldstein, A., Dioun, C., Santibáñez Godoy, D. G., & Osley-Thomas, R. (2016). The financialization of US higher education. Socio-Economic Review, 14(3), 507–535.
- Eichelberger, B., Mattioli, H., & Foxhoven, R. (2017). Uncovering barriers to financial capability: Underrepresented Students' access to financial resources. *Journal of Student Financial Aid*, 47, 3.

- El-Gamal, M. A. (2003). Interest and the paradox of contemporary Islamic law and finance. Fordham International Law Journal, 27(1), 108–149.
- El-Gamal, M. A. (2009). Islamic finance: Law, economics, and practice. Cambridge University Press.
- Esson, J., & Ertl, H. (2016). No point worrying? Potential undergraduates, study-related debt, and the financial allure of higher education. Studies in Higher Education, 41(7), 1265–1280.
- Ettlinger, N. (2004). Toward a critical theory of untidy geographies: The spatiality of emotions in consumption and production. *Feminist Economics*, 10, 21–54.
- Etzioni, A. (1987). On thoughtless rationality (rules-of-thumb). Kyklos, 40(4), 496-514.
- Findlay, J., & Hermannsson, K. (2019). Social origin and the financial feasibility of going to university: The role of wage penalties and Availability of funding. *Studies in Higher Education*, 44(11), 2025–2040.
- Fornell, P. H., & Larcker, D. F. (1981). Evaluating structural equation models with unobservable variables and measurement error. *Journal of Marketing Research*, 18(1), 39–50.
- Franke, N., & Piller, F. (2004). Value creation by toolkits for user innovation and design: The case of the watch market. *Journal of Product Innovation Management*, 21(6), 401–415.
- Furlong, A., & Biggart, A. (1999). Socio-economic disadvantage and access to higher education, initial findings of work from development Programme on young people. University of Glasgow.
- Galadima, W. M. (2015). Islamic home finance in the United Kingdom. Origin Press.
- Gefen, D., Straub, D., & Boudreau, M.-C. (2000). Structural equation modeling and regression: Guidelines for research practice. Communications of the Association for Information Systems, 4(7), 1–79.
- Ghaffari, A., & Çiftçi, A. (2010). Religiosity and self-esteem of Muslim immigrants to the United States: The moderating role of perceived discrimination. *International Journal for the Psychology of Religion*, 20(1), 14–25.
- Glick, P., Sakallı-Uğurlu, N., Akbaş, G., Orta, I. M., & Ceylan, S. (2016). Why do women endorse honor beliefs? Ambivalent sexism and religiosity as predictors. *Sex Roles*, 75(11–12), 543–554.
- Guest, M., Scott-Baumann, A., Cheruvallil-Contractor, S., Naguib, S., Phoenix, A., Lee, Y., & Al Baghal, T. (2020). Islam and Muslims on UK University campuses: Perceptions and challenges. Durham University, London: SOAS, Coventry: Coventry University and Lancaster: Lancaster University.
- Hair, J. F., Matthews, L. M., Matthews, R. L., & Sarstedt, M. (2017). PLS-SEM or CB-SEM: Updated guidelines on which method to use. *International Journal of Multivariate Data Analysis*, 1(2), 107–123.
- Hair, J. F., Ringle, C., & Sarstedt, M. (2011). PLS-SEM: Indeed, a silver bullet. The Journal of Marketing Theory and Practice, 19(2), 139–152.
- Hair, J. F., Ringle, C. M., Ringle, C. M., & Mena, J. A. (2012). An assessment of the use of partial least squares structural equation modeling in marketing research. *Journal of the Academy of Marketing Science*, 40(3), 414–433.
- Hair, J. F., Jr., Sarstedt, M., Hopkins, L., & Kuppelwieser, G. V. (2014). Partial least squares structural equation modeling (PLS-SEM) an emerging tool in business research. *European Business Review*, 26(2), 106–121.
- Hammond, L. (2013a). Somali transnational activism and integration in the UK: Mutually supporting strategies. *Journal of Ethnic and Migration Studies*, 39(6), 1001–1017.
- Hammond, L. (2013b). History, overview, trends and issues in major Somali refugee displacements in the near region (Djibouti, Ethiopia, Kenya, Uganda and Yemen). *Bildhaan: An International Journal of Somali Studies*, 13, 55–79.
- Hanson, S. (2006). Thinking back, thinking ahead: Some questions for economic geographers. In S. Bagchi-Sen &
 H. Lawton Smith (Eds.), *Economic geography: Past, present and future*. Routledge.
- Harris, H. (2004). The Somali community in the UK: What we know and how we know it. Information Centre about asylum and refugees in the UK (ICAR).
- Hassan, F., Musse, M., Jama, J., & Mohamed, F. (2013). *Mapping of the Somali diaspora in England and Wales*. International Organization for Migration.
- HEA. (2017). Key facts and figures: Higher education 2015/16. HEA Retrieved from http://hea.ie/assets/ uploads/2017/06/HEA-Key-Facts-and-Figures-201516.pdf
- Henseler, J., Hubona, G., & Ray, P. A. (2016). Using PLS path modeling in new technology research: Updated guidelines. *Industrial Management and Data Systems*, 116(1), 2–20.
- Hogg, M. A. (1996). Social identity, self-categorization, and the small group. In E. H. E. Witte & J. H. Davis (Eds.), Understanding group behavior (Vol. 2). Psychology Press.
- Hogg, M. A. (2016). Social identity theory. In S. McKeown, R. Haji, & N. Ferguson (Eds.), Understanding peace and conflict through social identity theory. Springer International Publishing.
- Holman, C., & Holman, H. (2003). First steps in a new country: Baseline indicators for the Somali community in LB hackney. Sahil Housing Association.
- Hulland, J. (1999). Use of partial least squares (PLS) in strategic management research: A review of four recent studies. Strategic Management Journal, 20(2), 195–204.
- Ibn Jibreen. (2006). Rule of Student Loan with Riba in Islam. Retrieved from www.ibn-jebreen.com/fatwa/vmasal-6087-.html. Accessed October 21, 2019.
- Ind, N., Fuller, C., & Trevail, C. (2012). Brand together: How co-creation generates innovation and re-energizes brands. Kogan Page Publishers.

Islamic Figh Academy in India. (2010). Getting loan for education: Islamic perspective. IFA Publication.

- Ismail, I. (2010). Legal stratagems (Hiyal) and usury in Islamic commercial law by Muhammed Imran Ismail. A Ph.D. thesis submitted to The University of Birmingham Department of Theology and Religion College of Arts and Law the University of Birmingham. Available at: http://etheses.bham.ac.uk/id/eprint/1325. Accessed December 1, 2021.
- Ismail, R. (2009). Financing education: Are there alternatives to students loan? Seminar Ekonomi, 9, 1–12.
- Ismail, S., Bakri, M. H., Ali, R., & Noor, A. M. (2014). Developing a framework of Islamic student loan-backed securitization. *Procedia*—Social and Behavioral Sciences, 129, 380–387.
- Jacewicz, E., Fox, R. A., O'Neill, C., & Salmons, J. (2009). Articulation rate across dialect, age, and gender. *Language Variation and Change*, 21(2), 233–256.
- Jones, T. P., Ram, M., & Theodorakopoulos, N. (2010). Transnationalism as a force for ethnic minority Enterprise? The case of Somalis in Leicester. *International Journal of Urban and Regional Research*, 34(3), 565–585.
- Joreskog, K. G. (1971). Statistical analysis of sets of congeneric tests. Psychometrika, 36(2), 109–133.
- Jungblut, J., Vukasovic, M., & Steinhardt, I. (2020). Higher education policy dynamics in turbulent times—access to higher education for refugees in Europe. Studies in Higher Education, 45(2), 327–338.
- Kahin, M. (1997). Educating Somali children in Britain. Trentham.
- Kant, I. (1785/1989). The foundations of the metaphysics of morals. Prentice-Hall.
- Kant, I. (1787/1997). The critique of practical reason. Cambridge University Press.
- Karbhari, Y., Naser, K. H. M., & Shahin, Z. (2004). Problems and challenges facing the Islamic banking system in the west: The case of the UK. *Thunderbird International Business Review*, 46(5), 521–543.
- Kena, G., Musu-Gillette, L., Robinson, J., Wang, X., Rathbun, A., Zhang, J., Wilkinson-Flicker, S., Barmer, A., & Dunlop Velez, E. (2015). *The condition of education 2015*. U.S. Department of Education, National Center for Education Statistics.
- Khaleel, F., & Avdukić, A. (2020). Exploring the epistemology of utility function in Islamic economics: Foundation of Islamic Finance (23-40). In V. Efendic (Ed.), *Islamic finance practices experiences from South Eastern Europe*. The Palgrave Macmillan.
- Kincheloe, J. L., & Steinberg, S. R. (2007). Cutting class: Socioeconomic status and education. Rowman & Littlefield Pub.
- Klaczynski, P. A. (2000). Motivated scientific reasoning biases, epistemological beliefs, and theory polarization: A two-process approach to adolescent cognition. *Child Development*, 71, 1347–1366.
- Klaczynski, P. A., & Fauth, J. (1997). Developmental differences in memory-based intrusions and self-serving statistical reasoning biases. *Merrill–Palmer Quarterly*, 43, 539–566.
- Klaczynski, P. A., & Gordon, D. H. (1996). Self-serving influences on adolescents' evaluations of belief-rel- evant evidence. Journal of Experimental Child Psychology, 62, 317–339.
- Klaczynski, P. A., & Lavallee, K. L. (2005). Domain-specific identity, epistemic regulation, and intellectual ability as predictors of belief-biased reasoning: A dual-process perspective. *Journal of Experimental Child Psychology*, 92(1), 1–24.
- Klucharev, V., Hytonen, K., Rijpkema, M., Smidts, A., & Fernandez, G. (2009). Reinforcement learning signal predicts social conformity. *Neuron*, 61(1), 40–51.
- Knowles, J. (1999). *Pity the poor students: Student funding and its impact on aspirations to higher education.* University of Humberside.
- Kohlberg, L. (1969). Stage and sequence: The cognitive-developmental approach to socialization. In D. Goslin (Ed.), *Moral development and behavior*. Holt, Reinhart, & Winston.
- Kramer, R. M., & Brewer, M. B. (1984). Effects of group identity on resource use in a simulated commons dilemma. Journal of Personality and Social Psychology, 46, 1044–1057.
- Kundu, P., & Cummins, D. D. (2013). Morality and conformity: The Asch paradigm applied to moral decisions. Social Influence, 8(4), 268–279.
- Kushner, T. (2006). Remembering refugees: Then and now. Manchester University Press.
- Kyambi, S. (2005). Beyond black and white: Mapping new immigrant communities. Institute for Public Policy Research.
- Laldin, M. A., & Furqani, H. (2015). Shariah scholars, ijtihad and decision making in Islamic finance. *Journal of Islamic Business and Management*, 5(1), 37–50.
- Lamborn, S. D., Dornbusch, S. M., & Steinberg, L. (1996). Ethnicity and community context as moderators of the relations between family decision making and adolescent adjustment. *Child Development*, 67(2), 283–301.
- Lee, J., & Mueller, J. A. (2014). Student loan debt literacy: A comparison of first-generation and continuing-generation college students. *Journal of College Student Development*, 55(7), 714–719.
- Lewis, I. (2017). Social and cultural anthropology in perspective: Their relevance in the modern world. Taylor and Francis.
- Loonin, D., & Morgan, J. (2019). Aiming higher: Looking beyond completion to restore the promise of higher education. Journal of Law and Education, 48(4), 423–448.

30

BERJ

- Mallinson, D. J., & Hatemi, P. K. (2018). The effects of information and social conformity on opinion change. PLoS ONE, 13(5), 1–22.
- Maloutas, T. (2007). Middle class education strategies and residential segregation in Athens. *Journal of Education Policy*, 22(1), 49–68.
- Maselko, J., & Kubzansky, L. D. (2006). Gender differences in religious practices, spiritual experiences and health: Results from the US general social survey. Social Science and Medicine, 62(11), 2848–2860.
- McCoy, S., & Smyth, E. (2011). Higher education expansion and differentiation in the Republic of Ireland. *Higher Education*, 61(3), 243–260.
- McDonough, P. M., & Calderone, S. (2006). The meaning of money: Perceptual differences between college counselors and low-income families about college costs and financial aid. *American Behavioral Scientist*, 49(12), 1703–1718.
- McDowell, L. (2005). Love and money: Critical reflections on welfare-to-work policies in the UK. *Journal of Economic Geography*, 5, 365–379.
- Metcalf, H. (1997). Class and higher education: The participation of young people from lower social classes. Council for Industry and Higher Education.
- Mews, C. J., & Abraham, I. (2007). Usury and just compensation: Religious and financial ethics in historical perspective. Journal of Business Ethics, 72(1), 1–15.
- Middlemiss, L. (2017). A critical analysis of the new politics of fuel poverty in England. *Critical Social Policy*, 37(3), 425–443.
- Miller, B. (1992). Collective action and rational choice place, community, and the limits to individual self-interest. *Economic Geography*, 68, 22–42.
- Mladenovic, R., Martinov-Bennie, N., & Bell, A. (2017). Business Students' insights into their development of ethical decision-making. *Journal of Business Ethics*, 155(1), 275–287.
- Moret, J. (2016). Cross-border mobility, transnationality and ethnicity as resources: European Somalis' post-migration mobility practices. Journal of Ethnic and Migration Studies, 42(9), 1455–1472.
- Morris, Y. (2016). Fulfilling our potential: Teaching excellence, social mobility and student choice. Consultation Response from CILIP Retrieved from https://doi.org/10.1017/CBO9781107415324.004. Accessed June 11, 2022
- Muhamad, N. (2008). *Muslim consumers' motivation towards Islam and their cognitive processing of performing taboo behaviors*. University of Western Australia.
- Mukhametzaripov, I. A., & Kozlov, V. E. (2017). The problem of shariah norms functioning in the legal system of a modern secular state. *European Research Studies Journal*, 20, 46–54.
- Mulcahy, C. M., Parry, D. C., & Glover, T. D. (2010). Play-group politics: A critical social capital exploration of exclusion and conformity in mothers groups. *Leisure Studies*, 29(1), 3–27.
- Office for National Statistics (ONS). (2020). Student loans in the public sector finances: A methodological guide. Office for National Statistics.
- Open Society Initiative for Europe; At Home in Europe & Open Society Institute. (2015). Somalis in European cities. Open Society Foundations.
- Opwis, F. (2007). Islamic law and legal change: The concept of Maslaha in classical and contemporary Islamic legal theory. Stanford University Press.
- Peet, R. (2000). Culture, imaginary, and rationality in regional economic development. *Environment and Planning*, *A*(32), 1215–1234.
- Petty, R. E., & Cacioppo, J. T. (1986). Communication and persuasion: Central and peripheral routes to attitude change. Springer-Verlag.
- Pollard, J., Datta, K., James, A., & Akli, Q. (2015). Islamic charitable infrastructure and giving in East London: Everyday economic-development geographies in practice. *Journal of Economic Geography*, *16*(4), 871–896.
- Pollard, N., Kantartzis, S., Ismail, M. M., Fransen-Jaïbi, H., & Viana-Moldes, I. (2019). The occupation of accessing healthcare and processes of (dis) citizenship in UK Somali migrants: Sheffield case study. World Federation of Occupational Therapists Bulletin, 75(1), 27–33.
- Prentice, D. A. (2008). Mobilizing and weakening peer influence as mechanisms for changing behaviour. In M. J. Prinstein & K. A. Dodge (Eds.), *Understanding peer influence in children andadolescents*. The Guilford Press.
- Rahman, F. (2009). *Major themes of the Qur'ān. Chicago*. The University of Chicago Press.
- Rahman, F. (2010). Revival and reform in islam : a study of islamic fundamentalism. Oneworld.
- Rasmussen, J. (2011). Education for Somali students in London: Challenges and strategies. DigitalCommons@ Macalester College.
- Reid, A. E., Carey, K. B., Merrill, J. E., & Carey, M. P. (2015). Social network influences on initiation and maintenance of reduced drinking among college students. *Journal of Consulting and Clinical Psychology*, 83, 36–44.
- Reid, A. E., Field, M., Jones, A., DiLemma, L. C., & Robinson, E. (2019). Social modelling of health behaviours: Testing self-affirmation as a conformity-reduction strategy. *British Journal of Health Psychology*, 24(3), 651–667.
- Richter, N. F., Sinkovics, R. R., Ringle, C. M., & Schlägel, C. (2016). A critical look at the use of SEM in international business research. *International Marketing Review*, 33(3), 376–404.

- Rigdon, E. E. (2012). Rethinking partial least squares path modeling: in praise of simple methods. Long Range Planning, 45(6), 341–358.
- Rigdon, E. E. (2014). Rethinking partial least squares path modeling: Breaking chains and forging ahead. Long Range Planning, 47(3), 161–167.
- Robinson, M. (2010). Conflict, child protection and religious communities: A review and recommendations on enhancing protection through partnership. Religions for Peace and UNICEF.
- Rowlingson, K., & Mckay, S. (2017). Financial inclusion annual monitoring report 2017. School of Social Policy, Birmingham Business School.
- Safian, Y. (2010). Necessity (Darura) in Islamic law: A study with special reference to the harm reduction Programme in Malaysia. LAP LAMBERT Academic Publishing.
- Salmi, J., & D'Addio, A. (2021). Policies for achieving inclusion in higher education. Policy Reviews in Higher Education, 5(1), 47–72.
- Sarstedt, M., Ringle, C. M., & Hair, J. F. (2017). Partial least squares structural equation modeling. In C. Homburg, et al. (Eds.), Handbook of market research (pp. 1–40). Springer, Cham.
- Sarstedt, M., Ringle, C. M., Smith, D., Reams, R., & Hair, J. F. (2014). Partial least squares structural equation modeling (PLS-SEM): A useful tool for family business researchers. *Journal of Family Business Strategy*, 5(1), 105–115.
- Sawyer, B., & Carroll, L. (2000). Attitudes to student financial support systems, systems 3 social research. In Annex N research report (Vol. II). Cubie Report.
- Scuzzarello, S., & Carlson, B. (2019). Young Somalis' social identity in Sweden and Britain: The interplay of group dynamics, socio-political environments, and transnational ties in social identification processes. *Migration Studies*, 7(4), 433–454.
- Searle, E., Jackson, L., & Scott, M. (2018). Widening access to classics in the UK: How the impact, public engagement, outreach and knowledge exchange agenda have helped. In A. Holmes-Henderson, S. Hunt, & M. Musié (Eds.), *Forward with classics: Classical languages in schools and communities*. Bloomsbury Academic, Bloomsbury Publishing Plc.
- Shaffer, M. (2018). Social control in transnational families: Somali women and dignity in Johannesburg. In *Gender* and mobility in Africa. Palgrave Macmillan, Cham.
- Sherif, M., Harvey, O.J., White, B.J., Hood, W.R., & Sherif, C.W. (1954/1961). Intergroup conflict and cooperation: The robbers cave experiment. Institute of Group Relations.
- Sianou-Kyrgiou, E., & Tsiplakides, I. (2011). Similar performance, but different choices: Social class and higher education choice in Greece. Studies in Higher Education, 36(1), 89–102.
- Siddiqi, M. N. (2004). What went wrong? Keynote address at the roundtable on Islamic economics: Current state of knowledge and development of discipline, held at Jeddah, Saudi Arabia on 26–27 may 2004 under joint auspices of the Islamic research and training institute, Jeddah; and the Arab planning institute, Kuwait. Retrieved from http://www.siddiqi.com/mns/Keynote_May2004_Jeddah.html. Accessed December 13, 2021.
- Siddiqui, A. S. (2005). Understanding and eliminating Riba: Can Islamic financial instruments be meaningfully implemented? *Journal of Management and Social Sciences*, 1(2), 187–203.
- Sierra, J. J., & McQuitty, S. (2007). Attitudes and emotions as determinants of nostalgia purchases: An application of social identity theory. *Journal of Marketing Theory and Practice*, 15(2), 99–112.
- Smith, S. J., Munro, M., & Christie, H. (2006). Performing (housing) markets. Urban Studies, 43, 81-98.
- Souaiaia, A. (2014). Theories and practices of Islamic finance and exchange Laws: Poverty of interest. *International Journal of Business and Social Science*, 5(12), 34–52.
- Stevenson, J. (2014). Internationalisation and religious inclusion in UK higher education. *Higher Education Quarterly*, 68(1), 46–64.
- Stevenson, J., & Baker, S. (2018). Widening participation to higher education Systems in Settlement Countries. In Refugees in higher education (great debates in higher education) (pp. 35–50). Emerald Publishing Limited. https://doi.org/10.1108/978-1-78743-714-220181004
- Stoet, G., & Geary, D. C. (2020). Gender differences in the pathways to higher education. Proceedings of the National Academy of Sciences, 117(25), 14073–14076.
- Strauss, K. (2008). Re-engaging with rationality in economic geography: Behavioural approaches and the importance of context in decision-making. *Journal of Economic Geography*, 8, 137–156.
- Sunley, P. (2000). Pension exclusion in grey capitalism: Mapping the pensions gap in Britain. Transactions of the institute of British Geographers, New Series, 25, 483–501.
- Sunstein, C. R. (2019). Conformity: The power of social influences. NYU Press.
- Syam, N. B., & Pazgal, A. (2013). Co-creation with production externalities. Marketing Science, 32(5), 805-820.
- Tameme, M., & Asutay, M. (2012). An empirical inquiry into marketing Islamic mortgages in the UK. International Journal of Bank Marketing, 30(3), 150–167. https://doi.org/10.1108/02652321211222531
- The Secretary of State for Business, Innovation and Skills. (2015). Fulfilling our potential: Teaching excellence, social mobility and student choice. Stationery Office.

Toder, N. L., & Marcia, J. E. (1973). Ego identity status and response to conformity pressure in college women. *Journal of Personality and Social Psychology*, 26(2), 287–294.

Turiel, E. (1983). The development of social knowledge: Morality and convention. Cambridge University Press.

- UK Parliament. (2022). Sharia-compliant alternative student finance. Retrieved from https://commonslibrary.parliament.uk/sharia-compliant-alternative-student-finance/. Accessed May 26, 2022.
- Ullah, S., Harwood, I. A., & Jamali, D. (2016). 'Fatwa repositioning': The hidden struggle for Shari'a compliance within Islamic financial institutions. *Journal of Business Ethics*, 17(2), 1–23.
- University Studies Council. (2015). Portland state university freshman year experience: A comprehensive review. University Studies Assessment Research, 11, Portland State University. Available at: https://pdxscholar.library. pdx.edu/cgi/viewcontent.cgi?article=1007&context=unst_assessment_research Accessed November 3, 2021.

Usmani, M. T. (2002). An introduction to Islamic finance. Kluwer Law International.

- Valentine, G., Sporton, D., & Nielsen, B. (2009). Identities and belonging: A study of Somali refugee and asylum seekers living in the UK and Denmark. *Environment and Planning: Society and Space*, 27, 234–250.
- Van Zanten, A. (2017). Widening participation in France and its effects on the field of élite higher education and on educational policy. In *Policy and inequality in education* (pp. 73–89). Springer, Singapore.
- Vertovec, S. (2004). Religion and diaspora. In P. Antes, A. W. Geertz, & R. R. Warne (Eds.), New approaches to the study of religion (Vol. 2). De Gruyter.
- Vitell, S. J., Paolillo, J. G., & Singh, J. J. (2005). Religiosity and consumer ethics. Journal of Business Ethics, 57(2), 175–181.

Vogel, F. E., & Hayes, S. L. (1998). Islamic law and finance. Kluwer Law International.

- Vogel, F. E., & Hayes, S. L. (2006). Islamic law and finance: Religion, risk, and return. Brill.
- Vossensteyn, H., & De, J. U. (2006). Student financing in The Netherlands: A Behavioural economic perspective. *Higher Education Dynamics*, 14, 213–240.
- Wakefield, S. E., & Poland, B. (2005). Family, friend or foe? Critical reflections on the relevance and role of social capital in health promotion and community development. Social Science and Medicine, 60(12), 2819–2832.
- Wakeling, P., Hampden-Thompson, G., & Hancock, S. (2017). Is undergraduate debt an impediment to postgraduate enrolment in England? *British Educational Research Journal*, 43(6), 1149–1167.
- Walter, T., & Davie, G. (1998). The religiosity of women in the modern west. *British Journal of Sociology*, 49, 640–660.
- Weinschenk, A. C., Panagopoulos, C., Drabot, K., & van der Linden, S. (2018). Gender and social conformity: Do men and women respond differently to social pressure to vote? Social Influence, 13(2), 53–64.
- Wold, H. (1980). Model construction and evaluation when theoretical knowledge is scarce: Theory and application of PLS. In J. Kmenta & J. B. Ramsey (Eds.), *Evaluation of econometric models* (pp. 47–74). Academic Press.
- Yu, R., & Sun, S. (2013). To conform or not to conform: Spontaneous conformity diminishes the sensitivity to monetary outcomes. PLoS ONE, 8(5), e64530.

How to cite this article: Avdukic, A., Khaleel, F., Abdullah, A., & Brawe, A. H. (2022). Religion as a barrier to the use of student loans for higher education: A community-based participatory study with Somalis living in England. *British Educational Research Journal*, *00*, 1–35. https://doi.org/10.1002/berj.3847

34 | BERJ

APPENDIX 1: THE DISTRIBUTION OF INDICATORS WITH THE FIRST ORDER

Questions	Dimension	First order	
I have used/been using student loan arrangements	a ₁	Panel A: Attitude towards Student Ioan	
For my study official financial loans are very important	a ₂		
For studying, you will receive financial support from relatives and friends	a ₃		
Financial schemes from student loan companies are very important	a ₄		
You are familiar with <i>ribā</i> (usury) term	b ₁	Panel B: Religious values	
Interest-based loans are prohibited like usury	b ₂		
It is Islamically prohibited to take interest-based loan hence student loan?	b ₃	Values	
In student loan contract there is a breach of Islamic principles	b ₄		
All conditions of student loan beyond interest are compliant with Islamic law	b ₅		
Taking the student loan is primarily for paying fees	с ₁	Panel C:	
Taking the student loan is primarily for life maintenance	c ₂	Perception of the	
Taking a loan was essential to complete my education	c ₃	purpose	
In case of not taking a student loan your education will not be fulfilled	C ₄	of student loans	
Taking a student loan contributed to your financial inclusion	c ₅		
I am confident in the permissibility of student loans based on available <i>fatawa</i>	d ₁	Panel D: Influence	
I am not confident in the permissibility of student loans based on available <i>fatawa</i>	d ₂	of Shariah Scholars	
Permissibility of student loans is justifiable on the basis of <i>darura</i> (necessity)	d ₃		
Permissibility of student loans is justifiable on the basis of <i>maslaha</i> (social benefit)	d ₄		
Avoiding student loan schemes is pre-requisite for safeguarding the religion	d ₅		
The permissibility of student loan based on <i>darurarh</i> (necessity) is rejected	d ₆		
Providing an alternative to student loans is the duty of the Muslim community	e ₁	Panel E: Communal solutions based on communal ethos	
Shari'ah scholars are responsible to suggest such a community-based solution for a student loan	e ₂		
Providing Shari'ah-compliant student loans is essential for Somali student	e ₃		
<i>Zakah</i> charity funds should be used for student needs in England not abroad	e ₄		



Questions	Dimension	First order	
There is no difference between Islamic and conventional student loans	f ₁	Panel F: Preference	
Having Shari'ah scholars approve a student loan gives you confidence	f ₂	for Islamic finance	
More expensive Shari'ah-compliant student loans are preferable to conventional	f ₃	style student loans	
Based on my experience with student loans, I will encourage others to use it	f ₄		
It is the Somalin community's duty to establish charity for investing in our youth	f ₅		